From Department of Clinical Neuroscience Karolinska Institutet, Stockholm, Sweden

NONSUICIDAL SELF-INJURY IN ADOLESCENCE: ATTEMPTS TO IMPROVE AND DEVELOP TREATMENT

Olivia Ojala



Stockholm 2023

All previously published papers were reproduced with permission from the publisher. Published by Karolinska Institutet. Printed by Universitetsservice US-AB, 2023 © Olivia Ojala, 2023 ISBN 978-91-8017-056-7

Nonsuicidal Self-Injury in Adolescence: Attempts to Improve and Develop Treatment Thesis for Doctoral Degree (Ph.D.)

By

Olivia Ojala

The thesis will be defended in public at Samuelssonsalen, Karolinska Institutet, Tomtebodavägen 6, 171 65 Solna, on November 10, 2023, 9:00.

Principal Supervisor:

Professor Clara Hellner Karolinska Institutet Department of Clinical Neuroscience Centre for Psychiatry Research

Co-supervisor(s):

Associate Professor Johan Bjureberg Karolinska Institutet Department of Clinical Neuroscience Centre for Psychiatry Research

Associate Professor Ralf Kuja-Halkola Karolinska Institutet Department of Medical Epidemiology and Biostatistics

Opponent:

Professor Dennis Ougrin King's College London Institute of Psychiatry, Psychology and Neuroscience Department of Child & Adolescent Psychiatry

Queen Mary University of London WHO Collaborating Centre for Mental Health Services Development Youth Resilience Research Unit

Examination Board:

Professor Mia Ramklint Uppsala University Department of Medical Sciences Division of Psychiatry Division of Child and Adolescent Psychiatry

Professor Viktor Kaldo Karolinska Institutet Department of Clinical Neuroscience Centre for Psychiatry Research

Linnaeus University Department of Psychology Faculty of Health and Life Sciences

Professor Laura Korhonen Linköping University Department of Biomedical and Clinical Sciences Center for Social and Affective Neuroscience

To youth suffering from self-injury

Popular science summary of the thesis

Nonsuicidal self-injury (NSSI) refers to deliberate injury to one's body that causes damage to the skin without suicidal ideation (e.g., through cutting, scraping, or banging). While NSSI is a behavior, there is also a mental disorder specifically for NSSI (i.e., NSSI disorder). NSSI and NSSI disorder are common among youth and often cause suffering and difficulties at school, home, and among friends. There is also a risk of developing other mental disorders and attempting suicide. Hence, preventing and treating NSSI is essential. Most people engage in NSSI with the expectation of escaping difficult emotions. According to theory, improving the ability to regulate emotions can reduce NSSI, and learning other less harmful ways to regulate emotions is thus a relevant treatment goal. Access to effective treatment for NSSI among adolescents is limited, often due to restricted resources. Delivering therapy over the internet can increase access to treatment, but we know little about how such treatment is experienced and if it is efficacious for NSSI among adolescents.

We recently developed an NSSI treatment entitled Internet-delivered Emotion Regulation Therapy for Adolescents (IERITA). The goal of this thesis was to develop and evaluate IERITA (Studies I-III), as well as to improve our understanding of how symptoms in childhood are associated with self-injury (Study IV).

In Study I, eleven adolescents with NSSI disorder and nine parents were interviewed about their experiences of IERITA. Both adolescents and parents appreciated the support from the online therapist and the flexibility to work with treatment whenever and wherever they wanted. However, the flexibility of the treatment was perceived as stressful for some. Parents treasured having their specific information in an online parent program, and the adolescents accepted that their parents were involved. Finding other ways to regulate emotions helped the adolescents to function better and refrain from NSSI. Nevertheless, for some adolescents, learning new emotion regulation skills and using them in daily life, was challenging. Based on what we learned from the interviews and feedback we received, we revised IERITA before conducting Study II.

In Study II, we evaluated the efficacy of IERITA in a randomized clinical trial. In total, 166 adolescents with NSSI disorder and past-month NSSI were randomized to either (1) IERITA as an addition to usual care or (2) usual care only (without IERITA). Adding IERITA to usual care was associated with few adverse events and reduced NSSI by 82% compared to 47% in usual care only. In addition, receiving IERITA contributed to greater reductions in emotion regulation difficulties, other destructive behaviors (e.g., anger bursts, taking drugs, drinking alcohol), and problems functioning in school, at home, and with friends, compared to regular care only. The advantage of IERITA was maintained at the three-month follow-up. In addition, improving one's ability to regulate emotions was

connected to reduced NSSI during IERITA. In sum, adding IERITA to usual care can be valuable when treating NSSI among adolescents.

In Study III, we followed up on Study II, using the same sample, and explored if any patient characteristics (e.g., age, suicidality, sleep difficulties) were relevant for the treatment outcome. We did not identify any subgroups that IERITA seemed to work better or worse for. However, parental invalidation (i.e., judging, denying, or diminishing their adolescents' emotions) was associated with a worse outcome regardless of whether adolescents received the add-on of IERITA. Hence, parents with high levels of invalidation may need support to understand and express that their adolescent's emotions are understandable. Finally, adolescents with NSSI disorder and varying mental health issues can potentially benefit from IERITA.

In Study IV, we investigated how Attention–Deficit/Hyperactivity Disorder (ADHD) symptoms (impulsivity, inattention, and hyperactivity) in childhood were associated with NSSI and suicidal behavior. We used an existing data set with 391 twins at risk of neurodevelopmental disorders (e.g., ADHD, autism spectrum disorder) that were followed over three to six years. Childhood inattention was related to NSSI, while childhood impulsivity was related to suicidal behavior. Childhood hyperactivity was not associated with any of the self-injurious behaviors. In sum, impulsivity and inattention in childhood are valuable to pay attention to and reduce, also concerning risk for self-injury.

To conclude, IERITA can be an efficacious, appreciated, and flexible treatment for adolescents engaging in NSSI. However, some adolescents may need additional support if IERITA is perceived as distressing. Helping youth find other ways to regulate emotions by adding IERITA can bring meaningful effects for many with few adverse effects. It is crucial to investigate the usefulness of IERITA in additional studies to understand how IERITA can work in different settings, such as within health care services. IERITA can potentially increase access to effective treatment among a wide range of adolescents engaging in NSSI. To prevent adolescent self-injury, helping children with high levels of impulsivity and inattention to reduce these symptoms can be one part of the puzzle. When and how to best prevent self-injury (e.g., practice impulse control in emotional situations) are questions to consider. In the future, adolescents engaging in NSSI may be offered online treatment within regular care, and new methods to prevent self-injury can be evaluated and used.

Abstract

Nonsuicidal self-injury (NSSI) and NSSI disorder are common in adolescence and associated with adverse outcomes, including suicide attempts. Detecting vulnerable children and adolescents is essential to decrease the risk of initiation as well as repetition of self-injury. Most individuals engage in NSSI to regulate difficult emotions; to effectively treat NSSI, targeting emotion regulation can thus be of importance. Accessibility to effective treatment is a challenge for adolescents engaging in NSSI. Internet-delivered treatment can increase access to treatment, but little is known about how such treatment is experienced and if it is efficacious for adolescents with NSSI or NSSI disorder. This thesis aimed to develop and evaluate the efficacy of Internetdelivered Emotion Regulation Therapy for Adolescents (IERITA), as well as to improve our understanding of early correlates of self-injurious behaviors.

Study I was a qualitative study where eleven adolescents with NSSI disorder and nine parents were interviewed regarding their experience of IERITA. The results showed that receiving support from an online therapist was essential to the treatment. Parents appreciated having their separate intervention, and adolescents accepted this form of parental involvement. Self-responsibility and the flexibility of the treatment were appreciated but could be distressing. Implementing new emotion regulation skills was relevant but perceived as challenging for some. Nevertheless, improved emotion regulation ability was a mentioned effect of IERITA that improved functioning and could hinder NSSI. It was concluded that IERITA is acceptable and can be developed to facilitate learning and address potential distress inherent to the treatment format.

Based on the findings from Study I and feedback from clinicians and patient organizations, we revised IERITA before conducting Study II. Study II was a randomized clinical trial comparing IERTA adjunctive to Treatment as Usual (TAU) to TAU only. Adolescents (*N* = 166) between 13 to 17 years old with NSSI disorder and past month NSSI were included. The IERITA plus TAU group reduced the NSSI frequency by 82% on average from pre- to posttreatment, significantly larger than the 47% reduction in the TAU-only group (blinded clinician-rated). Statistically significant improvements favoring ERITA plus TAU were also observed on most secondary outcomes, including other destructive behavior, emotion regulation difficulties, and functional impairment. In addition, the advantages of IERITA were maintained at the three-month follow-up. In line with theory, improvements in emotion regulation difficulties mediated reductions in NSSI. In addition, adolescents reported few adverse events concerning IERITA. It was concluded that IERITA can potentially improve access to evidence-based treatment for adolescents engaging in NSSI.

In Study III, as a secondary analysis of Study II, we investigated for whom IERITA was more or less helpful. Age, emotion regulation difficulties, depressive symptoms, function

level, parental invalidation, prior NSSI frequency, sleep difficulties, and suicidality were measured before randomization and investigated as moderators and predictors of treatment effect (i.e., change in NSSI frequency). No significant moderation effects were found. High parental invalidation was predictive of a less favorable outcome regardless of treatment condition. It was concluded that there is preliminary support for the efficacy of IERITA for adolescents with NSSI disorder with varying patient characteristics. In addition, detecting and reducing parental invalidation is of importance in NSSI treatments for adolescents.

In Study IV, the focus shifted to childhood. We investigated the associations between Attention–Deficit/Hyperactivity Disorder (ADHD) symptoms and NSSI as well as suicidal behaviors in children at risk of neurodevelopmental disorders (*N* = 391). The data came from an ongoing longitudinal cohort study of twins. At age 9 or 12, parents reported degree of ADHD symptoms and at age 15 clinicians assessed self–injury. Significant associations were found between childhood impulsivity and suicidal behavior, and inattention and NSSI. However, childhood hyperactivity was not significantly or meaningfully associated with either outcome. It was concluded that screening for and treating childhood impulsivity and inattention can be valuable to prevent self–injurious behaviors.

In sum, IERITA can be an efficacious, acceptable, and flexible treatment for adolescents engaging in NSSI. Nevertheless, additional support can be needed for those experiencing distress inherent to the online format. IERITA can increase access to evidence-based treatments for NSSI among a wide range of adolescents with NSSI disorder. Addressing difficulties in emotion regulation by adding IERITA to TAU can contribute to meaningful change for many with few adverse effects. Additional sufficiently powered trials are needed to understand the utility of IERITA across studies and samples. Inattention and impulsivity during childhood can, at least partly, help us understand self-injurious behaviors' development. Further studies are needed to establish the temporal order and understand the pathways of these associations. The contributions from the included studies can, by extension, innovate how we treat NSSI among adolescents and generate new hypotheses on preventing self-injury.

List of scientific papers

- I. Simonsson, O., Engberg, H., Bjureberg, J., Ljótsson, B., Stensils, J., Sahlin, H., & Hellner, C. (2021). Experiences of an Online Treatment for Adolescents With Nonsuicidal Self-injury and Their Parents: Qualitative Study. JMIR Formative Research, 5(7), e17910.
- II. Bjureberg, J., Ojala, O., Hesser., H, Häbel, H., Sahlin, H., Gratz, K. L., Tull, M. T., Claesdotter Knutsson, E., Hedman–Lagerlöf, E., Ljótsson, B., & Hellner, C. (2023). Effect of Internet–Delivered Emotion Regulation Individual Therapy for Adolescents With Nonsuicidal Self–Injury Disorder: A Randomized Clinical Trial. JAMA Network Open, 6(7), e2322069.
- III. Ojala, O., Hesser, H., Gratz, K. L., Tull, M. T., Hedman-Lagerlöf, E., Sahlin, H., Ljótsson, B., Hellner, C., Bjureberg, J. (2023). Moderators and Predictors of Treatment Outcome Following Adjunctive Internet–Delivered Emotion Regulation Individual Therapy Relative to Treatment as Usual Alone for Adolescents with Nonsuicidal Self–Injury Disorder. Unpublished manuscript.
- IV. Ojala, O., Kuja-Halkola, R., Bjureberg, J., Ohlis, A., Cederlöf, M., Norén Selinus, E., Lichtenstein, P., Larsson, H., Lundström, S., & Hellner, C. (2022). Associations of Impulsivity, Hyperactivity, and Inattention with Nonsuicidal Self-injury and Suicidal Behavior: Longitudinal Cohort Study Following Children at Risk for Neurodevelopmental Disorders into Midadolescence. BMC Psychiatry, 22(1), 679.

Contents

1	Literature review					
	1.1	Nonsuicidal Self-injury		3		
		1.1.1	Definition and classification	3		
		1.1.2	Nonsuicidal self-injury disorder	5		
		1.1.3	Developmental course and longitudinal outcome	6		
		1.1.4	Prevalence	6		
		1.1.5	Etiology and maintenance	7		
		1.1.6	Clinical correlates	10		
	1.2	Treatment for nonsuicidal self-injury		12		
		1.2.1 Pharmacological treatment		12		
		1.2.2	Psychosocial treatment	12		
		1.2.3	Mediation and moderation	14		
	1.3	Summ	nary	15		
2	Research aims					
3	The empirical studies					
	3.1	Treatment				
	3.2	Study I				
		3.2.1	Methods	22		
		3.2.2	Main results	22		
	3.3	Study II		23		
		3.3.1	Methods	23		
		3.3.2	Main results	25		
	3.4	Study III		26		
		3.4.1	Methods	26		
		3.4.2	Main results	26		
	3.5	Study IV		26		
		3.5.1	Methods	26		
		3.5.2	Main results	27		
	3.6	Ethical considerations		27		
4	Discussion					
	4.1	Main findings				
	4.2	Is IERITA an efficacious treatment for adolescents with NSSI disorder?				
	4.3	Can emotion regulation be a mechanism of change in IERITA?				
	4.4	What can we learn from the participant experience of IERITA?				
	4.5	For whom can IERITA be more or less helpful?				
	4.6	Can impulsivity and inattention in childhood be of importance to NSSI				
		and su	uicidal behavior in adolescence?	36		
	4.7	General limitations				

5	Conclusions			
6 Points of perspective			41	
	6.1	Replication	41	
	6.2	Stepped care	41	
	6.3	Implementation	42	
	6.4	Prevention	. 43	
7	Acknowledgements			
8	References			

List of abbreviations

ADHD	Attention-Deficit/Hyperactivity Disorder
A-TAC	Autism-Tics, ADHD, and other Comorbidities inventory
BPD	Borderline Personality Disorder
CATSS	Child and Adolescent Twin Study in Sweden
CBT	Cognitive Behavioral Therapy
CI	Confidence Interval
DBT	Dialectical Behavior Therapy
DBT-A	Dialectical Behavior Therapy for Adolescents
DSHI-Y	Deliberate Self Harm Inventory – Youth version
DSM-5	Diagnostic and Statistic manual of Mental disorders, fifth version
EAM	Experiential Avoidance Model
ERGT	Emotion Regulation Group Therapy
ERITA	Emotion Regulation Individual Therapy for Adolescents
IERITA	Internet-delivered Emotion Regulation Individual Therapy for Adolescents
IAPT	Improving Access to Psychological Therapies
IRR	Incidence Rate Ratio
K-SADS-PL	Schedule for Affective Disorders and Schizophrenia for School-Age Children-Present and Lifetime Version
NSSI	Nonsuicidal Self-Injury
OR	Odds Ratio
RCT	Randomized Clinical Trial
SD	Standard Deviation
SE	Standard Error
SSRI	Selective Serotonin Reuptake Inhibitors
TAU	Treatment as Usual

Introduction

Adolescence is a challenging time for many where a non-negligible group develops and suffers from nonsuicidal self-injury and the many adverse effects of it. Much has been done over the past years to increase our understanding of NSSI and to develop treatments for NSSI among adolescents. Still, the research field of self-injury is young, and several key questions remain. More knowledge is pressing as NSSI increases among our teenagers and is associated with suicide attempts, which is predictive of suicide – a leading cause of death among youth here in Sweden and other countries.

When studying to become a clinical psychologist, I volunteered to support adults and adolescents with self-injurious thoughts and behaviors through chat and telephone. With this experience of remote support and the target group, I started to wonder how internet-delivered treatment could work for self-injurious thoughts and behaviors. When it was time to write my master thesis in 2017, I encountered the IERITA project. At the time, two of my supervisors were conducting the first pilot study on IERITA. Once involved in the project, I never left. I started my clinical career with a complete focus on child and adolescent psychiatry. I was eager to spend my time building skills to treat and help this patient group with support from excellent clinicians. Many devoted and skillful researchers and clinicians have made these studies possible. And not to forget, all the hard-working adolescents and parents committed to doing their best to change their current situation. It has been an honor working as a therapist and project coordinator in IERITA, supporting youth to overcome their difficulties and to live a life that is meaningful to them.

The anticipation with the work within this thesis is that it can expand the possibilities for how we can treat NSSI among adolescents. Also, to bring insight into how to detect vulnerable children and prevent self-injurious behaviors. I am, nevertheless, humble towards the fact that this is just the start of things. I hope this work can inspire and encourage research involving vulnerable groups because much more is needed within adolescent self-injury.

)liviaO

Lerum, autumn, 2023

1 Literature review

1.1 Nonsuicidal Self-injury

1.1.1 Definition and classification

Self-injury without the intent to die is nothing new, and many terms have been used to describe the behavior over time. Descriptions of this behavior were found already in the mid-19th-century clinical records. The need for defined terminology of the phenomena stems from a long history of trying to help individuals struggling with such behavior.¹ Over the years, several names of the behavior have been suggested, and usual terms seen in the literature are, for example, deliberate self-harm, self-mutilation, self-cutting, and parasuicide. Different terms with slightly different definitions can potentially cause confusion among researchers and clinicians.² Today, the term nonsuicidal self-injury (NSSI) is commonly used. NSSI is defined as "the deliberate, self-directed damage of body tissue without suicidal intent and for purposes not socially or culturally sanctioned" (para. 1).³ This excludes accidental injury and socially acceptable behaviors such as tattooing and piercing. Common behaviors include cutting, banging/hitting, scratching, carving, and scraping.⁴ The intent of self-injury is not always easy to assess. Therefore, a conservative assessment is advised: all "non-zero", or vague evidence of suicidal intent should be assessed as suicidal self-injury, not NSSI.⁵

The concept of NSSI is debated⁶ and has been for a long time. The concerns put forward mainly concern two aspects. First, if it is meaningful to separate self-injury based on the intention; NSSI and suicidal behavior often co-occur, and assessing intention is difficult. Second, the classification of tissue damage particularly excludes cases of nonsuicidal self-poisoning. Although the term NSSI is debated and puts nonsuicidal self-poisoning in the diagnostic wilderness, NSSI is widely used in the literature and is acknowledged in the *Diagnostic and Statistical Manual of Mental Disorders, fifth version* (DSM-5).⁷ Still, self-injurious behaviors as a broader concept is also commonly used and could entail both NSSI and suicidal behavior, as illustrated in Figure 1.



Figure 1. Illustration of the relations between self-destructive and self-injurious behaviors and the suggested NSSI disorder

NSSI could also be classified as a self-destructive behavior, as illustrated in Figure 1. Selfdestructive behaviors, such as alcohol/substance abuse and disordered eating, can have deliberate self-harmning intentions without tissue damage; if so, they can be classified as indirect self-injury.^{8,9} As illustrated in Figure 2, behaviors can vary on a spectrum of suicidal and self-harming intention,¹⁰ and tissue damage. Assessing these intentions and tissue damage can guide intervention for a given individual.



Figure 2. Illustration of how self-destructive behaviors can have different intentions and degree of tissue damage

Within psychiatric disorders, self-injury is one of nine criteria of borderline personality disorder (BPD). At least five criteria need to be presented to fulfill BPD. NSSI is commonly endorsed in those with BPD; around 80% of those fulfilling BPD have engaged in NSSI in the past year.^{11,12} However, not all adolescents engaging in NSSI fulfill the criteria for BPD; in previous studies, 21–52% of adolescents engaging in NSSI seem to fulfill the criteria for BPD.¹³⁻¹⁶ Hence, classifying NSSI solely within BPD seems insufficient. In some countries, a psychiatric disorder is needed for health care or insurance to fund interventions; such structure could increase the risk of misclassifying those with NSSI as having BPD or excluding those with NSSI without comorbidities from treatment.¹⁷ With this in mind, we will shift the focus from NSSI, the *behavior*, to the *disorder*.

1.1.2 Nonsuicidal self-injury disorder

One pressing question is how to capture those needing NSSI treatment and if a separate disorder can offer guidance. Several researchers have put forward the suggestion of a separate disorder for NSSI since the 1980s.^{218,19} In 2013, NSSI disorder was included in the latest diagnostic manual, DSM-5, as a disorder for further study. In short, the proposed criteria entail that NSSI (A) is present on five or more days the past year, (B) is expected to regulate emotions and/or interpersonal conflicts, (C) is preceded by inter- and intrapersonal difficulties and/or preoccupation of the behavior is present and hard to control, (D) is not socially sanctioned (e.g., piercing), (E) causes significant distress or impaired function, and (F) is not better explained by another disorder or condition.⁷

As a disorder for further study, more research is needed to validate the criteria and investigate the clinical utility. Although ten years have passed since NSSI disorder was introduced, few efforts have been made to validate the suggested disorder.²⁰ However, the studies that have been conducted indicate that NSSI disorder is distinct from other disorders and connected to adverse events beyond comorbidity.^{14,21,22} Nevertheless, concerns have been raised, for instance, that the frequency criterion is too low²¹⁻²³ and that the distress/impairment criterion is difficult to assess.¹⁰ Validating and potentially establishing NSSI disorder can contribute to a consistent definition of clinically significant NSSI. Defining clinically significant NSSI can facilitate communication in research and practice, assessment of NSSI, and development and investigation of treatments for NSSI.^{20,24}

Since NSSI disorder is not yet acknowledged as an established disorder, less research has focused on NSSI disorder compared to NSSI (behavior). In the upcoming sections, literature concerning the *behavior* of NSSI, will be mentioned as "NSSI". When literature concerns the *disorder*, it will be mentioned as "NSSI disorder". In addition, in cases where there is no distinction regarding suicidal intention, "self-injury" will be used.

1.1.3 Developmental course and longitudinal outcome

Knowing the prevalence, age of onset, and developmental course is critical in detecting and treating NSSI. Generally, NSSI is most common during adolescence.²⁵ The probability of engaging in NSSI increases exponentially from age 9 to 10; the most common age for first NSSI engagement is around age 13 to 14.^{26,27} For many, NSSI or self-injury presented in adolescence decreases or ceases in adulthood.²⁸⁻³⁰ From what is known so far, those continuing with NSSI over adolescence or young adulthood seem to have more depressive symptoms and emotion regulation difficulties compared to those who ceased NSSI.³¹³² The developmental course of NSSI can involve setbacks and relapses^{30,33} and switch to other self-destructive behavior;^{30,34} the course of NSSI is thus complex.

Although the majority ceases to engage in NSSI over time, there is reason to take NSSI in adolescence seriously. Sporadic and recurrent NSSI during adolescence is related to an increased risk of incidence of psychiatric disorders in later adolescence.³⁵ Furthermore, engaging in NSSI during adolescence is also connected to an elevated risk of adverse outcomes in adulthood, including substance abuse, symptoms of or diagnosed psychiatric disorder, using psychotropics, and poorer employment outcomes.^{28,36} The risks are even higher for those engaging in NSSI and suicidal self-injury.^{28,36} NSSI is predictive of future suicide attempts, which in turn is predictive of suicide death.^{37,38} Among those aged 15 – 19, suicide is the leading cause of death in Sweden and the leading or second cause of death in many countries in Europe and North America.³⁹ Hence, preventing and treating self-injury is an urgent call. It is encouraging that the risks can be reduced if NSSI ceases, although it may take time.^{30,40} Hence, early preventive and interventive strategies for NSSI are important.

1.1.4 Prevalence

How prevalent NSSI is depends on what time frame one considers, how one asks (i.e., checklist or open question), and in which population.^{41,42} It should be said that NSSI is a stigmatized behavior, and the decision to disclose NSSI is complex.⁴³ In the general population, the pooled lifetime prevalence has been estimated to be 5.5% among adults, 13.4% among young adults, and 17.2% among adolescents;⁴¹ in another more recent meta-analysis, the lifetime prevalence for adolescents was estimated to be 22.9%, and past-year prevalence 18.6%.⁴⁴ Less is known about clinical samples of adolescents; the knowledge is based on relatively small samples. It is reasonable to believe that 30–60% of adolescents within healthcare have ever or in the past year engaged in NSSI.^{14,45-47} Regarding NSSI disorder, the estimated prevalence is 3.1–7.6%^{48,49} for adolescents and 0.8% for adults.⁵⁰ Thus, if NSSI disorder becomes an established disorder, it could be a common psychiatric disorder among adolescents; in comparison, depressive or anxiety disorders have estimated prevalence rates of 2.6% and 6.5%, respectively⁵¹. In summary, independent of how it is measured, NSSI is present in adults and adolescents, with a

higher prevalence among adolescents. Childhood and adolescence could be optimal times to prevent and treat NSSI.

Regarding gender differences in the prevalence of NSSI, the results have been inconsistent. In a comprehensive meta-analysis,⁵² it was concluded that females were more likely to report lifetime NSSI than males, with an odds ratio of 1.50 (95% CI = 1.35, 1.65). Furthermore, the gender difference was more prominent for clinical and younger samples. However, sample selection might influence the results as men are less likely to seek psychiatric treatment and, therefore, are underrepresented in clinical samples. ^{52,53} NSSI disorder seems also to be more prevalent among females than males.⁴ Gender minority groups have been studied to a lesser extent. Recent studies have shown elevated prevalence of NSSI in gender minority groups compared to cisgender groups.^{54,55} The critical point here is to remember that NSSI and NSSI disorder seem to be present in all studied gender groups.

The prevalence of NSSI has increased over time. Studies exploring temporal trends until 2015, showed an increase in both lifetime and recurrent NSSI among adolescents, adults, and across genders.^{56,57} The greatest increase was found in girls and young women.^{56,58} Between 2014 and 2021, one study showed an increased prevalence of lifetime engagements in NSSI among adolescents.⁵⁹ Some possible reasons for the increases include, first, that younger generations could be more exposed to NSSI through media/internet, which could potentially impact contagion between individuals.^{60,61} Second, the stigmatization of NSSI could be decreasing,⁵⁶ resulting in more disclosures. In fact, there seems to be a general increase in mental health issues among the young, with uncertain causes.⁶² Yet, this is an important area for future research as it can inform health care services and society on what to target.

1.1.5 Etiology and maintenance

1.1.5.1 Risk factors

Our ability to predict NSSI is weak. However, this is not unique to NSSI but representative of self-injurious thoughts and behaviors.⁶³ We know more about correlates (i.e., factors associated with another factor) of NSSI than risk factors (i.e., correlates that precede the outcome). From where we stand today, we have limited knowledge of how single factors or combination of factors can constitute a risk for NSSI repetition and onset.⁶⁴⁻⁶⁶ In a meta-analysis published in 2015, the strongest predictors for repeating NSSI were a prior history of NSSI, cluster b personality disorders (e.g., BPD), and hopelessness.⁶⁵ Some of these risk factors may be shared with other behaviors; for instance, prior history of NSSI is also a risk factor for suicidal behaviors.⁶³ There is some evidence for heritability of NSSI of around 40–60%,⁶⁷ but no specific genes have been found associated with NSSI. The etiology and maintenance of NSSI are presumably influenced

by social, psychological, and biological factors and their interactions. Less attention has been given to biological factors. Recently, studies have shown some evidence regarding alterations in the stress system, specific brain regions involved in regulating emotions, and decreased pain sensitivity⁶⁸ concerning NSSI. Since NSSI most often initiates in adolescence, understanding early risk factors is vital to understand the development and prevent initiation.

1.1.5.2 Maintaining factors

A usual question in society is why some engage in NSSI. A common misconception is that it is just "attention-seeking,"⁶⁹ fostering unhelpful behaviors and stigmatization. From a functional approach (i.e., assuming that behaviors are driven by their instant antecedents and consequences), NSSI is maintained because it fulfills crucial functions for the individual short-term. The different functions/motivators of NSSI are usually separated into intrapersonal (e.g., reduce unpleasant thoughts or feelings) or social/interpersonal (e.g., escape difficult social situations or facilitate access to help).^{70,71} Functions could differ between individuals, but also over time, and NSSI could fulfill several functions simultaneously.^{71,72} Engaging in NSSI to regulate emotions is particularly common.⁷¹ However, social functions are likely underreported because intrapersonal functions could be seen as more socially acceptable.⁷³ Understanding why and how NSSI is maintained is essential knowledge for optimizing psychosocial treatments for NSSI.

1.1.5.3 Theoretical models

Several theoretical models have been put forward to describe the etiology and maintenance of NSSI based on the available empirical base at the time. In this thesis, the focus is on models with a functional approach, and such models are common in the literature. In 2006, Chapman, Gratz, and Brown proposed a model, the Experiential Avoidance Model (EAM)⁷⁴ of self-injury, that has been widely spread. This model proposes that NSSI is primarily a strategy for reducing or escaping unwanted emotional arousal. As illustrated in Figure 3, how one acts in a situation evoking emotional response can be understood from the interaction of several factors (e.g., emotion intensity, poor distress tolerance); these factors are results of previous experiences and learning. The presence of any or several of these factors can lead to experiential avoidance: choosing NSSI to avoid or escape from unwanted internal experiences. This temporary desired emotional relief makes NSSI more likely to occur in similar situations in the future. However, avoiding emotional experiences can have paradoxical effects; emotional avoidance can contribute to more frequent and more intense emotions, which can trigger NSSI. The vicious cycle of NSSI can be exacerbated over time as NSSI can be linked with fewer punishing consequences (such as decreased pain and fear) and knowledge through experience that NSSI can be a way to decrease emotional distress

quickly. Some promising psychosocial treatments for NSSI are built upon this model, which we will learn more about in later sections.



THE EXPERIENTIAL AVOIDANCE MODEL (EAM) OF DELIBERATE SELF-HARM

Figure 3. Graphical illustration of the steps in the Experiential Avoidance Model (EAM). In: Chapman, A.L., Gratz, K.L., & Brown, M.Z. (2006) Solving the puzzle of deliberate self-harm: The experiential avoidance model. Behaviour Research and Therapy, 44, 371-394. Reprinted with permission from Elsevier.

Other models have also had a great impact on the research field, for instance, the integrated model by Nock in 2009.⁶⁴ In extension to the EAM model, the integrated model proposes that distal risk factors, such as genetic predisposition, childhood abuse, and familial hostility, all create a disposition of having difficulties regulating one's internal experiences or environment. This vulnerability can, in a stressful situation, together with the presence of hypothesized vulnerability factors for NSSI (e.g., seeing or knowing others engage in NSSI, desire for self-punishment), lead to NSSI. In a more recent model, the Benefits and Barrier Model by Hooley and Franklin, published in 2018,⁷⁵ benefits of and barriers to NSSI are described to explain why some engage in NSSI while others do not. In their model, several benefits of NSSI are explained, such as affiliating to a group, self-punishment, means of communication, and emotional regulatory processes. In addition, several barriers are stated that discourage NSSI engagements, such as aversion to NSSI and pain, that can be targeted to hinder NSSI. In summary, although not all steps

of these models have been sufficiently empirically tested, they offer several hypotheses of preventing and treating NSSI worth considering further.

1.1.5.4 Emotion regulation

Many theoretical models include the notion that NSSI can regulate emotional experiences. In a broad sense, emotion regulation is the process of monitoring, evaluating, and modifying emotional reactions.⁷⁶ It is a multidimensional construct where several definitions exist. A commonly used definition, stated by Gratz and Roemer in 2004, conceptualizes emotion regulation as "(a) awareness and understanding of emotions, (b) acceptance of emotions, (c) ability to control impulsive behaviors and behave in accordance with desired goals when experiencing negative emotions, and (d) ability to use situationally appropriate emotion regulation strategies flexibly to modulate emotional responses as desired in order to meet individual goals and situational demands" (p.42).⁷⁷ Thus, many different abilities could be helpful in successful emotion regulation. Emotion dysregulation (i.e., difficulties with emotion regulation) is common across many psychiatric disorders, including depressive, anxiety, substance abuse disorders.^{78,79}

There are several ways to regulate emotions. Depending on the situation, individuals can use both adaptive and maladaptive strategies. The choice of using NSSI specifically to regulate emotions can be influenced by, for example, (1) pragmatic aspects as it could be rapid, effective, and easily implemented; (2) having seen or heard others using NSSI for this reason; and (3) other behaviors not having fulfilled the expected goal.⁶⁴ Potentially, NSSI has stronger neurobiological effects⁶⁸ than other maladaptive emotion regulation strategies. For instance, when the pain from tissue damage during NSSI relives, this could increase positive affect and decrease negative affect and physiological arousal.⁷⁵ Studying the mechanism of emotion regulation during NSSI could be informative regarding how treatments can be optimized.

1.1.6 Clinical correlates

There is a wide range of clinical correlates to NSSI and NSSI disorder, including disorders such as depressive, anxiety, disruptive, substance use, and posttraumatic stress disorder.^{13,14,46} In this section, we will go into depth with some clinical correlates closely related to NSSI. They can offer ways to understand the developmental pathway of self-injurious behaviors.

1.1.6.1 Suicidal behavior

Suicidal behavior should always be considered in the light of NSSI. Suicidal behavior refers to directly self-injurious behaviors (e.g., suicide attempt or suicide) with some level of intent to end one's life.⁵ Although the motivator for suicidal behavior is to end one's life, emotion regulation difficulties are also associated with suicidal behavior.^{80,81}

This may not be surprising given the overlap between NSSI and suicidal behavior. The magnitude of overlap can vary, with 19–70% of adolescents with NSSI or NSSI disorder reporting at least one suicide attempt.^{13,45,46,82-84} How and why NSSI is linked to suicidal behavior is not fully understood, and several models can offer partial explanations. The three most preeminent theories will be described in short and more extensive summaries, including supporting evidence, can be found elsewhere.^{85,86} The *Third variable* theory suggests that NSSI and suicidal behavior are linked because they are associated with a third variable (e.g., depression). The *Gateway* theory proposes that self-injurious behavior, such as suicide attempts. The *Acquired capability* theory (stemming from Joiner's interpersonal theory of suicide⁸⁷) suggests that the fear of pain, injury, and death associated with a suicide attempt can decrease as individuals habituate toward pain and injury through engaging in NSSI. Based on these theories, treating NSSI, or shared risk factors for NSSI and suicidal behavior, can be potential ways of decreasing the risk of self-injurious behaviors.

1.1.6.2 Neurodevelopmental disorders

Neurodevelopmental disorders are conditions that have an onset during childhood and are characterized by cognitive and motor functioning impairments.⁷ One of the most common neurodevelopmental disorders is Attention–Deficit/Hyperactivity Disorder (ADHD).⁸⁸ ADHD is the neurodevelopmental disorder that has been studied the most concerning self-injury. Although other neurodevelopmental disorders may be understudied regarding self-injury, there is reason to believe that ADHD is especially relevant to NSSI and suicidal behavior.

ADHD subdimensions. Emotion dysregulation, a central factor to NSSI and a correlate to suicidal behavior, is expected within ADHD.⁸⁹ This may partly explain why associations between ADHD and self-injury⁹⁰⁻⁹² have been found in systematic reviews. On a more detailed level, how the separate ADHD subdimensions (i.e., impulsivity, hyperactivity, and inattention) relate to self-injury is not fully understood. Symptoms of these subdimensions are common in children and not exclusive to ADHD.^{93,94} Impulsivity, or impulsivity and hyperactivity combined have, in many studies, shown associations with NSSI and suicidal behavior.⁹⁵⁻⁹⁸ However, the effect of hyperactivity has rarely been investigated separately, and the relative contribution of hyperactivity is uncertain. Turning to inattention, the few existing studies concerning self-injury have shown mixed results.^{95,96,99} A more detailed understanding of impulsivity, hyperactivity, inattention, and other risk factors or correlates (e.g., emotion dysregulation), can hopefully contribute to better preventive and intervention strategies for self-injury.

1.2 Treatment for nonsuicidal self-injury

At the end of the 20th century, the first randomized clinical trials investigating treatment targeting self-injury among youth were published. Within the past ten years, several researchers have reviewed¹⁰⁰⁻¹⁰⁴ or evaluated through meta-analyses¹⁰⁵⁻¹⁰⁷ the evidence of treatment for self-injury among youth. Some of these publications consider the evidence of treatment for NSSI specifically.¹⁰²⁻¹⁰⁴ The current knowledge is primarily based on small samples, including girls. In the upcoming sections, pharmacological and psychosocial treatment will be described separately.

1.2.1 Pharmacological treatment

The effect of pharmacological treatment is rarely investigated for NSSI specifically. Currently, no pharmacological treatment can be seen as established for treating NSSI.¹⁰⁸ In clinical guidelines, pharmacological treatment for self-injury is not recommended; it is only indicated for comorbidity.^{109,10} A proportion of adolescents with self-injury are treated with psychotropics, as thus can be an indicated treatment for comorbid disorders.⁵⁸ However, little is known about how these medications affect self-injury specifically, making it critical to investigate this further. For adolescents with a history of NSSI, pharmacological treatment with antidepressants may be associated with a risk that NSSI can persist or re-occur over the treatment period.^{45,11} How such results could generalize to other comorbid disorder or in combination with psychosocial treatment targeting NSSI remains unclear.

1.2.2 Psychosocial treatment

Several psychosocial treatments are promising for NSSI, including mentalization-based therapy, Cognitive Behavioral Therapy (CBT), and Dialectical Behavior Therapy for Adolescents (DBT-A).^{101-103,105} DBT-A is an evidence-based treatment for self-injury and NSSI evaluated in independent randomized clinical trials.¹⁰³ However, some researchers recommend further evaluations of DBT-A to become more confident of the generalizability of the results.¹⁰⁰ Although DBT-A can be effective in treating NSSI among adolescents, it is an extensive treatment where accessibility is limited.¹¹² Briefer treatments could be valuable complements, offered early in the development of NSSI. There are some novel and promising brief treatments for adolescent NSSI; for instance, the eight to twelve-session-long *Cutting Down Programme* based on CBT and DBT.¹¹³ However, additional studies are needed to establish the efficacy of brief treatments targeting NSSI among adolescents.

1.2.2.1 Emotion regulation individual therapy for adolescents

In 2006, Emotion Regulation Group Therapy (ERGT) was developed with the hope of being an economically and clinically feasible treatment for adults with Borderline Personality Disorder (BPD) and NSSI.¹¹⁴ ERGT is an acceptance-based behavioral therapy

founded on the theoretical framework of the Emotional Avoidance Model (EAM).⁷⁴ In ERGT, emotion regulation skills are practiced with the expectation that reducing emotion regulation difficulties will decrease NSSI. ERGT, as a brief add-on treatment, is effective in decreasing NSSI, improving emotion regulation ability, and other health outcomes among women with BPD.^{15,16} Subsequently, ERGT was adapted into an adolescent version, a treatment named Emotion Regulation Individual Therapy for Adolescents (ERITA).¹¹⁷ In contrast to ERGT, ERITA is an individual therapy where a parent is included and participates in a separate online parent course. In 2017, the first study on ERITA was published; in an uncontrolled pilot trial including adolescents with NSSI disorder (N = 17),¹¹⁷ the results showed significant improvements corresponding to a 42% reduction in NSSI episodes from pre- to posttreatment, with further improvement at the 6-month follow-up. The results were also promising regarding treatment adherence and credibility. Hence, these brief treatments focusing on emotion regulation seem promising in treating NSSI among adults and adolescents.

1.2.2.2 Internet-delivered treatment

Internet-delivered treatment (also known as online, computerized, digital, or webbased) has, over the past 20 years, shown to be effective for several mental health disorders for adults and adolescents.^{118,119} The most common form of internet-delivered treatment is reading self-help material online guided by an online therapist for a predetermined time frame. Some of the potential advantages of internet-delivered treatment include saving therapist time, large-scale access, allowing patients to work at their own pace where it suits them, saving traveling time, less drift from treatment manual (as material is structured), continuous monitoring of outcome, and easier access to the therapist.¹²⁰ Nevertheless, the online format presents several challenges, for instance, a risk of missing out on clinical information, technical issues, and responsibility to engage in the treatment according to plan.^{120,121} It is clear that this format offers advantages that could be of importance to patients and organizations, but not without challenges.

Internet-delivered treatment for self-injurious behaviors is less explored than other symptoms or disorders. One potential reason for this is that therapists have concerns on how to respond to elevated suicide risk through the online format.¹²² Hence, handling elevated suicide risk has to be carefully considered in the remote format (e.g., offering telephone or face-to-face contact if needed). However, more youth with self-injury could potentially access psychosocial treatment with this format; in fact, some might prefer the online format (or be reluctant or refuse traditional face-to-face contact)¹²³ and the access to care has shown to be unequal for adolescents with self-injury depending on geographical localization (at least in the United Kingdom).⁵⁸ In a review published in 2019 on internet- and mobile-delivered treatment for self-injury,¹²⁴ the knowledge base was scarce, especially for adolescent self-injury treatments; there

were two studies investigating supplementary mobile-delivered skills training for adolescents with self-injury and one pilot study on internet-delivered treatment for adolescents with NSSI disorder namely Internet-delivered ERITA (IERITA). The latter study on IERITA was conducted within our research group.

1.2.2.3 Internet-delivered emotion regulation individual therapy for adolescents

From the face-to-face manual of ERITA, an internet-delivered version was developed and tested in two pilot trials. In the first uncontrolled pilot trial published in 2018 (N = 25),¹²⁵ the results showed comparable results concerning utility, acceptability, and feasibility to the open pilot trial of face-to-face ERITA.¹¹⁷ In a recent randomized feasibility trial (N = 30)¹²⁶ in Denmark, the results showed that investigating IERITA as an addition to treatment as usual (TAU) compared to TAU only was feasible. Hence, IERITA seems to be an acceptable and feasible treatment for adolescents with NSSI and NSSI disorder. A large-scale randomized clinical trial is motivated to investigate efficacy compared to a control condition.

1.2.2.4 Experience of online treatment

The efficacy of any treatment is highly dependent on its acceptability. Letting adolescents with self-injury share their treatment experience and be involved in the development of treatments has been requested.¹⁰⁰ For internet-delivered treatment targeting self-injury, more knowledge on, for instance, the impact of technical issues and how the therapeutic alliance is experienced has been articulated.¹²⁴ Qualitative studies (such as interviewing participants and analyzing the transcriptions) contribute to in-depth knowledge of experience that could be valuable together with quantitative studies to develop treatment.

1.2.3 Mediation and moderation

One area for advancements within psychotherapy research in general¹²⁷ is understanding how and why a particular treatment works. When we detect a treatment effect, it leaves us wondering why we see this effect. Studying mediators (i.e., variables that may account for the effect between treatment and outcome¹²⁷) can be a starting point to answer the question. Emotion regulation difficulties have been shown to mediate change in NSSI in the small uncontrolled pilot trials of face-to-face ERITA¹¹⁷ and IERITA.¹²⁵ Investigating the mediating effect of emotion regulation in IERITA relative to other treatments without this theoretical foundation can provide evidence of specificity in treatment over and above nonspecific factors.

Another aspect we need to understand more about within NSSI¹⁰⁸ and self-injury¹⁰³ is treatment moderation: for whom and under what circumstances a specific treatment works.¹²⁸ Investigating nonspecific treatment predictors (i.e., variables associated with

treatment outcome regardless of treatment condition¹²⁸) is more common and can be evaluated in uncontrolled study designs. Studying treatment moderators and predictors can inform about the utility of treatment and be helpful in treatment decisions.

1.3 Summary

Nonsuicidal self-injury (NSSI) is prevalent among adolescents and is a risk factor for suicide attempts predictive of suicide, a common reason for death among youth in several countries. Accessible, evidence-based treatments for NSSI and NSSI disorder are highly needed. Delivering treatment online increases access to treatment, but the efficacy of internet-delivered treatment for NSSI is unknown. Emotion regulation is a common function of NSSI, and targeting emotion regulation in treatment can potentially reduce NSSI. A brief, Internet-delivered Emotion Regulation Individual Therapy for Adolescents (IERITA) seems promising regarding results from small pilot studies. Investigating the experience of IERITA more in-depth can increase our understanding of how IERITA works and can further be improved. A large-scale randomized clinical trial is essential to understand the efficacy of IERITA (compared to a control condition). Such a design also allows for more specific evaluations of whom the treatment works for and if the effect of IERITA is mediated via the theorized process of emotion regulation. At the same time, understanding the development of and preventing self-injury is also crucial. ADHD symptoms are common in childhood and related to both emotion regulation and self-injury with and without suicidal intention. Studying the effects of impulsivity, hyperactivity, and inattention in childhood on NSSI and suicidal behavior in adolescence can hopefully improve preventive and intervention strategies.

2 Research aims

The overarching aim of this thesis was to develop and evaluate Internet-delivered Emotion Regulation Individual Therapy for Adolescents (IERITA) with Nonsuicidal Self-Injury (NSSI) disorder and to improve our understanding of the early correlates of selfinjurious behaviors.

Study IObjective: In participants from the pilot study of IERITA, we wanted to
explore the experiences of IERITA among adolescents and their parents.
We aimed to investigate how they experienced the online format,
treatment content, the model with separated modules for adolescents and
parents, and potential perceived changes in daily life.

Hypothesis: The study was exploratory, and no hypothesis was stated beforehand.

Study IIObjective: To investigate the efficacy and the potential mediating effect of
emotion regulation difficulties of IERITA adjunctive to Treatment as Usual
(TAU) compared to TAU only.

Hypothesis: Those randomized to IERITA adjunctive to TAU would show larger decreases in NSSI frequency (primary outcome), emotion regulation difficulties, other destructive behaviors, psychiatric symptoms, and functional impairment (secondary outcomes) after treatment termination compared to those randomized to TAU only. Furthermore, we hypothesized that the decrease in NSSI would be mediated by a decrease in emotion regulation difficulties for those randomized to IERITA adjunctive to TAU.

Study IIIObjective: Following Study II, we wanted to understand for whom IERITA
works better or worse. Eight variables were investigated as moderators and
predictors based on theoretical and empirical justifications.

Hypothesis: Given the novelty of IERITA, potential moderators and predictors were exploratively investigated.

Study IV Objective: In a population-based twin cohort following children into midadolescence, we wanted to study the associations of childhood impulsivity, hyperactivity, and inattention with NSSI and suicidal behavior in children at risk for neurodevelopmental disorders (e.g., ADHD).

Hypothesis: Impulsivity would be associated with NSSI and suicidal behavior. Hyperactivity and inattention were investigated exploratively based on limited and conflicting previous knowledge.

3 The empirical studies

Most of the included studies in this thesis cover the development and evaluation of Internet-delivered Emotion Regulation Individual Therapy for Adolescents (IERITA; Studies I-III). The main methods and results will be summarized in this section. Please see the respective publication or manuscript for a more complete description of methods and results.

3.1 Treatment

Foundation of IERITA. IERITA was developed by a team of clinicians, researchers, webbased experts, and representatives of patient organizations for self-injury. The Child Internet Project in Stockholm, Sweden, hosted the web-based treatment platform used in Studies I-III. The interface of the platform was designed to be youth-friendly. Both adolescents, parents, and therapists log into the platform through a doubleauthentication process. The treatment modules resemble book chapters that the adolescent or parent can navigate by clicking through. The modules include audio files, videos, and interactive worksheets. Figure 4 shows examples of the interface of IERITA.



Figure 4. Screenshot from worksheets in Internet–delivered Emotion Regulation Individual Therapy for Adolescents (ERITA). In: Bjureberg, J., et al. (2018). Extending research on Emotion Regulation Individual Therapy for Adolescents (ERITA) with nonsuicidal self–injury disorder: open pilot trial and mediation analysis of a novel online version. BMC Psychiatry, 18(1), 1–13. Reproduced with permission. Adolescent treatment. The adolescent treatment consists of 11 modules delivered over 12 weeks. A new module is opened every week with the option of taking a one-week break if needed. In the treatment, adolescents follow fictional characters to facilitate learning (social modeling). The treatment starts with psychoeducation about NSSI and how it is associated with emotional avoidance. The difference between short-term gains and long-term negative consequences of NSSI is clarified to increase motivation to engage in treatment. In the second module, impulse control skills are introduced to offer concrete skills to refrain from NSSI early in treatment. Throughout the treatment, the focus is put on increasing well-being through acting according to valued directions. The following modules aim to increase skills in emotional awareness: to learn to label emotions and use emotional information in adaptive ways. In the first half of the treatment, mindfulness skills are introduced. In the fifth module, participants practice emotional acceptance: to stay with discomfort without trying to change or modify the experience, to be able to do what is important in life. Once these skills have been taught, emphasis is put on emotional approach: identifying, labeling, and investigating what information the emotion is carrying and considering acting according to such information. Emotional distraction is taught as a first step before approaching the emotion if needed. As practice in emotional approach and distraction continue, selfvalidation is added to decrease self-judgment. The treatment finishes with a summary and information about how to handle obstacles and setbacks, to consolidate knowledge and foster maintained effects.

Mobile application. Adolescents receive access to a complementary mobile application where they can (1) practice skills training; (2) access their crisis plan; (3) register daily presence or absence of NSSI (or other self-destructive behaviors) impulses or acts, and the factors preceding the impulses, as well as protective factors, to learn more about one's behavioral pattern; and (4) be reminded of their homework assignment for the week. When a new module opens, more information and functionalities are added to the app. The start page in the mobile app is shown in Figure 5.


Figure 5. Illustration of the mobile app from Internet-delivered Emotion Regulation Individual Therapy for Adolescents (IERITA). In: Bjureberg, J., et al. (2018). Extending research on Emotion Regulation Individual Therapy for Adolescents (ERITA) with nonsuicidal self-injury disorder: open pilot trial and mediation analysis of a novel online version. BMC Psychiatry, 18(1), 1–13. Reproduced with permission.

Parent course. The parent course consists of six modules that the parent can choose to work through over six or twelve weeks. Parents will learn skills to understand and support their adolescent and handle their own emotional reactions. The parent also gets access to the adolescent treatment material every week to support the adolescent's skills training. The parent course starts with psychoeducation about NSSI and emotional reactivity (i.e., the tendency to experience high emotional intensity) to better understand the symptoms the adolescent is experiencing. In addition, parents are introduced to practice in *emotional awareness* to understand and handle their own emotional reactions. Parents are taught *validation* (i.e., recognizing, understanding, and accepting other's emotional experiences) to decrease invalidation of their adolescent's emotional reactions. Furthermore, parents practice *self-validation* to decrease judgments of their emotional reactions. In the latter part of the course, *behavioral activation* is introduced to increase positive time together and time for likable or relaxing activities for the parent. The treatment finishes with a summary and information on how to handle obstacles and setbacks, to consolidate knowledge and foster maintained effects.

Therapist contact. The therapist contact is asynchronous; the therapist responds to the participant during weekdays office hours within two days. Both adolescents and parents have contact with an online therapist. The therapist gives feedback, explains content if needed, supports the generalization of skills, reminds the family of the treatment modules, and answers questions from adolescents and parents.

3.2 Study I

3.2.1 Methods

3.2.1.1 Study design, participants, and procedure

Adolescents and their parents who participated in the pilot study of IERITA¹²⁵ were invited to interviews after treatment termination. When selecting participants for interviews, the goal was to get as much variation as possible based on age, gender, and NSSI frequency pre- and posttreatment. In total, nine adolescents and eleven parents participated separately in interviews. The interviews were based on interview protocols with open-ended questions regarding the experience of IERITA. My colleague and I, at the time psychology master students, previously unknown to the participants, conducted and transcribed the interviews under supervision.

3.2.1.2 Data analysis

The analysis method was based on thematic analysis by Braun and Clarke,¹²⁹ a commonly used method for finding and analyzing patterns, codes, and themes within interview data. A data-driven approach was chosen, meaning that codes of interest were not decided before the analysis but rather emerging in the analysis process. We separated data from adolescents and parents to look at the convergence of the information from different sources and assess the consistency of findings (i.e., data source triangulation).

3.2.2 Main results

Three overarching themes were found. The theme "Support can come in different shapes" described how the participants received support. First, contact with the online therapist was an essential treatment feature. For some adolescents, communication was more effortless in the online format compared to face-to-face contact. Second, adolescents saw advantages in involving parents, such as having a common language. Third, the fictional characters one followed in the treatment and having access to a mobile app supported everyday skills training. The second theme "Self-responsibility can be empowering as well as distressing" described that the self-responsibility inherent to the online format could confer advantages and challenges. The high flexibility in time and place was appreciated, and being in charge of the treatment work created a sense of pride and empowerment. However, some were distressed because of the treatment, leading to difficult emotions and thoughts, procrastination, and avoidance. The third theme "Acquiring new skills and treatment effects" described the outcomes during and after IERITA in daily life. The analysis did not point to any emotion regulation skill of specific importance. Instead, there was great variation between individuals and over time in helpful skills. The experienced changes included improved communication and acceptance of emotions and needs and improved knowledge and ability to implement emotion regulation strategies, which in turn, for some were mentioned as reasons for decreased or ceased NSSI.

3.3 Study II

3.3.1 Methods

3.3.1.1 Study design, participants, and procedure

Study II was a randomized clinical trial comparing IERITA adjunctive to Treatment as Usual (TAU) to TAU only at three different sites in Sweden (Stockholm, Västra Götaland, and Skåne). Adolescents 13 to 17 years old with NSSI disorder and engaging in NSSI the past month were included. Interested families were either clinician or self-referred and screened for eligibility via telephone and face-to-face. Participants answered self-reports both before, during, and after treatment (post and three-month follow-up). In addition, adolescents and parents participated in blinded assessments by telephone after treatment termination and at the three-month follow-up.

3.3.1.2 Outcome measures

The primary outcome measure was NSSI frequency, as measured with the Deliberate Self Harm Inventory – Youth version (DSHI–Y^{130,131}), both self–reported and clinician–rated. Secondary outcomes included emotion regulation difficulties, other self–destructive behaviors, functional impairment, and psychological distress. Adverse events, treatment adherence, credibility, and satisfaction were also monitored for the IERITA plus TAU group.

3.3.1.3 Interventions

Development of IERITA. Based on the findings of Study I and the received feedback from therapists and patient organization representatives, we revised IERITA before conducting Study II. First, more emphasis was put on increasing dedication and commitment to using emotion regulation skills that could displace NSSI early in the treatment. Second, several adaptions were made to facilitate learning and skills training, including, adding a quiz about emotions, audio-recorded treatment summaries, and adding treatment summaries (both text and audio) in the app. Third, the section about valued direction was expanded and moved to the end of treatment. The reason for moving the section about valued direction was to focus primarily on displacing NSSI in the first modules. This priority order aligns with DBT-A,¹³² where decreasing life-threatening behaviors is the first priority. Increasing behaviors connected to well-being were introduced later when NSSI potentially had decreased. Fourth, questions about how the treatment was progressing and the need for additional support were added halfway through treatment. The anticipation with this revision was to normalize that treatment work can be challenging and detect the need for additional therapeutic support. Lastly, seeking situations that elicited emotional responses, both imaginary and in everyday life, to practice emotional approach was dropped. The experience was that it was challenging to identify appropriate situations for some participants. Instead, more focus was put on noting everyday risk situations (including emotions) for NSSI or other self-destructive behavior, generate suitable emotion regulation skills, and practice to access emotion regulation skills flexibly. Using emotion regulation skills flexibly is central in decreasing emotional avoidance and the risk of NSSI.⁷⁴

Therapists. The clinicians were psychologists or psychotherapists who worked parttime as therapists in the study. All therapists received education in the intervention, NSSI, and suicide assessment, as well as supervision during the trial.

Control condition. An ideal control condition should be potentially efficacious, safe, and desired.¹³³ At the time when we designed Study II, we landed in a TAU condition for the following reasons: (1) there were no evidence-based treatments targeting youth NSSI specifically at the time, and a TAU condition represented standard care, i.e., what is currently available to adolescents with NSSI, and; (2) given the potential suicidal risk and comorbidities, a strict waitlist control was not an option. As participants could have other treatment needs beyond NSSI, all participants were allowed to continue or initiate TAU. We also considered that discontinuing existing care could expose risks to the participants.¹³³ TAU was enhanced in the sense that weekly self-reports were monitored by study clinicians and the research team, and procedures according to safety protocols were followed if deterioration in health was detected.

3.3.1.4 Statistical analysis

The primary and secondary outcome measures were measured with mixed effects modeling. Mixed effects modeling (sometimes also referred to multilevel modeling, or growth models) was used based on the dependency between observations as participants were measured repeatedly, and missing values were assumed to be missing at random. To model change in the primary outcome, NSSI frequency, a zero-inflated negative binomial generalized mixed effects regression model was used. This model can be suitable when there are many zeros in the data and the non-negative counts (e.g., 1, 2, 3) follow a right-skewed distribution (e.g., many with low number of NSSI episodes and a long "tail" with few observations of many episodes).

The mediation analysis used a parallel process latent growth curve modeling approach for mediation.¹³⁴ Separate latent growth curves were performed for the hypothesized mediator (emotion regulation difficulties) and outcome (weekly NSSI frequency during and one-month posttreatment). Next, the two latent growth curves were combined to investigate if IERITA plus TAU, relative to TAU-only, influenced the growth (change) in emotion regulation difficulties that, in turn, affected the growth (change) in NSSI.

3.3.2 Main results

In total, 166 adolescents (93% girls, mean age = 15.0, SD = 1.2) participated in the trial; 84 families were randomized to IERITA plus TAU and 82 families to TAU only. Reduction in NSSI frequency was significantly larger in the IERITA plus TAU versus TAU-only for both self-reported NSSI (β [SE] = -0.08 [0.02]; *P* <.001; IRR = 0.29 [95% CI = 0.14-0.58]) and blinded clinician-rated NSSI (β [SE] = -1.09 [0.27]; *P* < .001; IRR = 0.34 [95% CI = 0.20-0.57]). From the month before to the one-month posttreatment, IERITA plus TAU had reduced NSSI frequency by 82%, whereas TAU-only had reduced by 47% (blinded clinician-rated). Furthermore, IERITA plus TAU improved significantly than TAU-only regarding emotion regulation difficulties, other self-destructive behaviors, and functional impairment. The significant superiority of IERITA plus TAU remained at the three-month follow-up. There was no significant group difference regarding psychiatric symptoms at one-month posttreatment or follow-up.

Participating in IERITA plus TAU, relative to TAU-only, was associated with greater reductions in emotion regulation difficulties, which led to greater reductions in NSSI frequency (indirect effect = -0.028 [95% bootstrap CI = -0.053 to -0.010]).

Adolescents completed on average 9.6 (SD = 2.2) of 11 modules, and parents completed 5.5 (SD = 1.0) of 6 modules. The treatment satisfaction rated by adolescents after IERITA was relatively high (mean = 25.2 out of 32, SD = 5.2). After completing the first module, adolescents rated IERITA as moderately credible (mean = 20.0 out of 27, SD = 4.1). The therapist time for IERITA (when therapists were reviewing treatment work, giving feedback, and answering questions) was on average 378.8 (SD = 163.1) minutes per family. Regarding adverse events, five participants (6%) in IERITA plus TAU and nine participants (11%) in TAU-only reported suicide attempts during the treatment period or at one-month posttreatment or three-month follow-up. In the IERITA plus TAU group 4 (5%) participants experienced adverse events connected to IERITA at one-month posttreatment. The events entailed increased sadness, increased stress, and increased self-destructive behaviors.

The type and frequency of TAU during the study did not differ significantly between the groups. In total, 118 participants (76%) received psychosocial treatment. The most common type and frequency was supportive therapy biweekly or monthly. Altogether, 67 participants (43%) received psychopharmacological treatment. Antidepressants,

specifically Selective Serotonin Reuptake Inhibitors (SSRI), was the most frequent medication.

3.4 Study III

3.4.1 Methods

3.4.1.1 Study design, participants, and procedure

Study III was a secondary analysis following Study II with the same setting, intervention, participants, and procedure.

3.4.1.2 Measures

The outcome was weekly self-reported NSSI frequency during and for four weeks after treatment termination measured with the DSHI-Y.^{130,131} The eight theoretically and empirically motivated variables to test as moderators and predictors were measured before randomization; the eight variables were age, emotion regulation difficulties, depressive symptoms, function level, parental invalidation, prior NSSI frequency, sleep difficulties, and suicidality.

3.4.1.3 Statistical analysis

A zero-inflated negative binomial generalized linear mixed effects regression model was used (same as in Study II). The client factor and its interactions with time were added to investigate prediction. Then, the client factor and its interactions with time and the treatment condition were added to investigate moderation.

3.4.2 Main results

No significant moderation effects were found. Parental invalidation significantly predicted the treatment effect (β [SE] = 0.027 [0.011]; *P* = .014). Thus, higher levels of parental invalidation before treatment started were predictive of less favorable outcomes regardless of treatment condition.

3.5 Study IV

3.5.1 Methods

3.5.1.1 Study design, participants, and procedure

Study IV was a longitudinal cohort study following twins at risk for neurodevelopmental disorders or common comorbidity from childhood into mid-adolescence. The data was collected from the Child and Adolescent Twin Study in Sweden (CATSS).¹³⁵ CATSS is an ongoing population-based cohort study investigating health (i.e., somatic, and mental) from childhood to adulthood. At age 9 or 12, a parent was interviewed via telephone regarding neurodevelopmental disorders symptoms. Families with children at risk for

neurodevelopmental disorder were invited to a face-to-face clinical assessment at age 15.

3.5.1.2 Measurements

The exposures in childhood were the degree of impulsivity, hyperactivity, and inattention as measured with the Autism–Tics, ADHD, and other Comorbidities inventory (A-TAC).^{136,137} The two outcomes in mid-adolescence were lifetime engagement of (1) NSSI and (2) suicidal behavior (yes or no) measured with the semi-structured clinical interview Schedule for Affective Disorders and Schizophrenia for School-Age Children-Present and Lifetime Version (K-SADS-PL),¹³⁸ the second version published in 2009.¹³⁹ We adjusted for age at childhood interview (9 or 12), birthyear, sex, and other neurodevelopmental disorder symptoms.

3.5.1.3 Statistical analysis

Odds ratios were calculated using logistic regression, applying generalized estimating equations, and cluster robust standard errors. Because observations between twins are clustered (non-independent), a method that does not make assumptions about independence between observations was needed. The outcomes were investigated in ten separate models. First, the three different exposures were investigated separately in univariable models with each outcome. Second, all exposures were included in the same model. Third, both exposures and potential confounders were included.

3.5.2 Main results

The sample (N = 391) included 225 (57.5%) boys and 166 (42.5%) girls. At age 15, 32 (8.2%) adolescents had ever engaged in NSSI, and 18 (4.6%) in suicidal behavior. The association between inattention and NSSI was statistically significant in all models (fully adjusted model: OR = 1.99, 95% CI = 1.36–2.91, p < 0.001). In addition, the association between impulsivity and suicidal behavior was statistically significant in all models (fully-adjusted model: OR = 2.08, 95% CI = 1.13–3.81, p = 0.018). The other associations (i.e., hyperactivity and NSSI or suicidal behavior, inattention and suicidal behavior, impulsivity and NSSI) were statistically non-significant in all models.

3.6 Ethical considerations

Ethical considerations are of great concern in all the steps of a scientific project and should weigh the risks against potential gains. Certain care should be taken regarding youth and those with potentially life-threatening behaviors. The studies in this thesis have received ethical approval, and several ethical considerations have been made.

First, the design of the treatment studies was carefully chosen to maximize safety. In Study II, being randomized to the one group that did not include internet-delivered treatment could disappoint families. However, the control group received IERITA after the three-month follow-up. Nevertheless, families were informed of the study design via telephone and face-to-face to minimize the potential disappointment. In Studies I-III, all participating adolescents were allowed, encouraged, and recommended regular care based on their needs; this was a precaution taken given the experimental nature of the intervention and the risk of not getting other interventions based on their care needs. All participants were registered in regular care during the treatment phase to facilitate rapid access to regular care. All adolescents were monitored weekly during the intervention period and at follow-up. Therapists and members of the research group monitored the answers on weekdays. In case of detected deterioration, the family was contacted according to a set procedure. Members of the research group, psychologists and a senior child and adolescent psychiatrist, were contactable on demand and participated in weekly treatment conferences. Altogether, these additional components (e.g., weekly monitoring and regular treatment conferences) contributed to enhanced care, compared to regular care only, for all participants.

Self-reports containing sensitive questions (e.g., suicidal behavior, NSSI) seem to have no iatrogenic effects on a group level,^{140,141} but could still be burdensome for some. The number of questions was kept to a minimum by choosing short, validated scales. Regular self-reports were discussed with the adolescent at the assessment meeting to detect potential adverse effects. In case of missed self-reports, we contacted the family according to a structured routine. This procedure aimed to detect any adverse effects of the questions or deterioration. When interviewing adolescents in Study I, we considered that talking about previous or current problems with somebody unknown could have a negative impact. We tried to minimize this by letting the adolescent decide on the format and place for the interview. The interview was guided by what adolescents wanted to talk about, and interviewers asked adolescents during and after the interview how it felt.

We considered that an experimental online treatment might not be suitable for the most severely affected. Adolescents with conditions in need of acute treatment (i.e., anorexia nervosa, psychosis, immediate suicide risk) were excluded from the treatment studies to focus on more extensive treatment at first hand. Experienced and trained clinicians tried to detect such severe conditions in both the telephone and face-to-face pre-assessment, to refer or recommend families to suitable care.

Participant safety and confidentiality in the online format were critical regarding the sensitivity of the data gathered. The platform was on a secure server where both therapists and participants logged in through a two-factor authentication process (i.e.,

both password and code) for both therapist and participants. Participants and therapists were automatically logged out in case of inactivity. The mobile app offered to adolescents was discreet with a neutral mobile app icon (e.g., not mentioning mental health) to protect adolescents' privacy. All participants were encouraged to engage with the treatment material in settings where they could be alone and undisturbed.

The choice of involving parents yields some ethical considerations concerning the autonomy of the adolescent. Some adolescents may not have wanted to participate in the treatment studies as at least one parent had to be informed and involved. If such concerns were communicated, therapists would discuss this with the adolescent and assist in communicating with the parent(s) if requested. From a safety and feasibility perspective, we assessed that there were potentially important positive effects of involving the parent(s). It could benefit the adolescent that the parent practiced skills to support the adolescent.^{103,104} Still, the integrity of the adolescent was important. It was necessary that adolescents had individual contact with the therapist online and during face-to-face assessment. Not the least to offer opportunities to disclose sensitive information. Also, when conducting the interviews in Study I, the interviews were conducted in separate rooms.

In comparison to the other studies, Study IV did not include a treatment-seeking group. Nevertheless, those potentially at risk of neurodevelopmental disorders were offered a professional assessment to better understand strengths and difficulties. Study IV included experienced clinicians who were observant of the status and aware of the sensitivity of the questions. They were prepared to give advice, recommend, or refer families if requested or needed.

In summary, although it can be challenging to handle risks when studying this target group, several steps can be taken to increase safety. Involving adolescents engaging in NSSI in studies evaluating treatments developed for them is vital for the results to be generalizable to the target group. Limited knowledge of effective treatments could have adverse effects through missing out on potentially effective treatments.

4 Discussion

4.1 Main findings

Internet-delivered Emotion Regulation Individual Therapy for Adolescents (IERITA) targeting Nonsuicidal Self-Injury (NSSI) was experienced as a treatment with high support levels where parents can be involved successfully. The flexibility inherent to the format was appreciated by many but challenging for some. Having access to emotion regulation skills was helpful to hinder NSSI, and several skills were beneficial to reduce emotion regulation difficulties. IERITA adjunctive to Treatment as Usual (TAU) was significantly more efficacious than TAU-only in reducing NSSI frequency, emotion regulation difficulties, other destructive behavior, and functional impairment. For the IERITA plus TAU group, reductions in emotional regulation difficulties were associated with reductions in NSSI. No significant moderators were found, but a high level of parental invalidation was predictive of less favorable outcomes irrespective of treatment condition. For children at risk of neurodevelopmental disorders, impulsivity was associated with suicidal behavior, while inattention was associated with NSSI.

4.2 Is IERITA an efficacious treatment for adolescents with NSSI disorder?

At this point, Study II offers the best opportunity to answer this question. What are the main findings that speak for the efficacy of IERITA? First, significantly larger reductions in self- and clinician-rated NSSI were found for IERITA plus TAU compared to TAU-only in a relatively large sample. The high module completion indicates the acceptability and relevance of IERITA. The moderate credibility and relatively high satisfaction suggest that IERITA can be appreciated by many interested in internet-delivered treatment. The external validity of the findings is strengthened by conducting the study within regular care and in a sample with a relatively high level of psychiatric comorbidity.

A subsequent question is *how* efficacious IERITA plus TAU is in relation to other psychosocial treatments targeting adolescent NSSI. Comparing results with previous treatment studies targeting NSSI among adolescents is challenging because of differences in: (1) how change in NSSI frequency is evaluated (e.g., proportion of individuals with NSSI 0, 1 to 3, 4–6, ≥7¹⁴² or 50% reduction in NSSI¹¹³ with varied timespans); (2) target groups (e.g., being at high risk of suicide, some degree of BPD criteria); (3) treatment length and intensity; (4) primary outcome (e.g., suicide attempts, NSSI thoughts, suicidal ideation). To contextualize the findings from Study II, results from a selection of previous treatment studies is summarized in Table 1. These studies are also randomized clinical trials with similar treatment lengths that evaluate self-injury frequency in a comparable timeframe. Still, as the target groups and sample sizes differ, I am hesitant to make direct comparisons between studies. Nevertheless, I believe some general patterns are worth commenting; overall, the magnitude of reduction in NSSI and average NSSI frequency after treatment for IERITA adjunctive to TAU are comparable to other psychosocial treatments with relatively similar treatment length. Moreover, the effects of IERITA adjunctive to TAU were accomplished with less therapist time on average, compared to several other treatments in Table 1 (where therapist time or session length was reported).

A considerable shortcoming within this research field is the lack of a threshold for clinically significant NSSI. To define treatment responders, using the absence of NSSI has been suggested by some.^{106,142} A significantly higher proportion of adolescents with no NSSI one-month posttreatment was observed in the IERITA plus TAU group (n = 43, 56%[8% missing]) compared to the TAU-only group (n = 26, 34% [6% missing]). However, this difference was non-significant at three-month follow-up (n = 33, 47% in IERITA plus TAU [15% missing], n = 33, 44% in TAU-only [9% missing]). From the studies presented in Table 1, only the study by Santamarina-Perez et al.¹⁴³ reported on NSSI absence. In this study, eleven adolescents (79% [21% missing]) receiving DBT-A reported no NSSI 13 to 16 weeks after treatment start. However, current or recent NSSI was not an inclusion criterion (as in Study II); thus, the proportion could, at least partly, be explained by the fact that some adolescents did not currently engage in NSSI when starting treatment. These observed values should be interpreted with caution given that missing values could greatly impact on the results (especially in small samples). If using the proportion of NSSI absence after treatment as a response rate, around 50% of responders to cognitive behavioral therapy is shared across several psychiatric disorders.¹⁴⁴

The average high-maintained reduction resulting in no or few episodes of NSSI indicates that IERITA adjunctive to TAU can have a clinically meaningful effect for many. Nevertheless, some may still need more or other treatment, and the effect remains unknown beyond three months after treatment. At this point, any additional reductions in NSSI compared to what is currently offered (that TAU could represent) are, in my opinion, meaningful.

Study	Sample	Treatment type	z	Treatment length and dose	Outcomes: frequency of NSSI or SI ^a
Bjureberg et al., 2023	Age 13-17, NSSI disorder, ≥ 1 NSSI episode in the past	IERITA + TAU	82	12 weeks 11 weekly treatment modules for adolescents and 6 for parents. Average weekly therapist time 32 to 34 minutes.	NSSI observed median week 13-16 = 0 (10R = 0-4, 8% missing). NSSI estimation week 13-16 = 1.2 (95% CI 0.9-1.4) ^b
(Study II)	month			TAU according to treatment need (typically supportive therapy biweekly)	82% estimated reduction in NSSI episodes from the month before to the month after treatment
				12 weekly group sessions (75 min)	SI (presented to emergency departments) observed mean week 12–24 = 0.2 (range 0–2)
Griffiths et al., 2019 ¹⁴⁵	Age 12-10, 2 1 31 episode In the past 6 months	MBT–Ai + TAU	26	TAU according to treatment need $^{\rm o}$	68% reduction from the 12 weeks before to the 12 weeks after treatment
	Are 12-17 > 5 NISSI enjoydes			8-16 weeks	NSSI observed median week 13-16 = 1 (IQR = 0-3)
Kaess et al., 2019 ¹¹³	the past 6 months, 2 1NSSI episode in the past month	Cutting down programme; CBT	37	8-12 sessions and parental sessions on as- needed basis (no info on sessions length)	90% reduction in NSSI episodes from the month before to the month after treatment
	Age 12-18, ک 2 previous SI, ک 1 دا میں 16 سرماری م 000	A TOO		19 weeks Wooddy individual accines (20 min) and	SI observed mean week 10-15 = 1.2 (SD = 2.0)
2014 ¹⁴⁶	criteria (or 3 if 2 were assessed as threshold)		30	weeky individual sessions (120 min) and multifamily sessions (120 min) with telephone and family sessions on an as-needed basis	Not possible to estimate reduction from before the treatment
Santamarina-	Age 12–17, NSSI disorder or			16 weeks Biweekly individual sessions (60 min), weekly	NSSI estimated mean week 13-16 = 1.3 (95% Cl 0.7- 1.9)
Perez et al., 2020 ¹⁴³	suicide attempt the past 12 months, and high risk of suicide	DBT-A	18	group sessions for adolescents (60 min), and separate for family (60 min) with telephone coaching on demand	Not possible to estimate reduction from before the treatment

Table 1. Nonsuicidal self-injury and self-injury outcomes after brief psychosocial treatment from a selection of randomized clinical trials targeting adolescent self-injury, experimental group only

Note. Only within-group results of the experimental condition are presented. BPD = borderline personality disorder, CBT = cognitive behavior therapy, CI = confidence interval, DBT-A = dialectical behavior therapy for adolescents, IERITA = internet-delivered emotion regulation individual therapy, IQR = interquartile range, MBT - Ai = mentalization based therapy introductory group manual for an adolescent population, NSSI = nonsuicidal self-injury, NSSI disorder = nonsuicidal self-injury disorder, RCT = randomized clinical trial, SI = self-injury, TAU = treatment as usual

^aResults presented here selected for comparison purposes to the results of Study II

^bExponentiated marginal predictions, blinded clinician-assessed

 $^\circ$ No information on the most common type and frequency of treatment as usual

It is encouraging that IERITA in adjunctive to TAU can also be efficacious for other symptoms. Significantly larger reductions for the IERITA plus TAU group compared to the TAU-only group were found for other destructive behaviors (such as drinking alcohol, taking drugs, anger bursts, binge eating), emotion regulation difficulties, and functional impairment. On the other hand, contrary to our hypothesis, there was no significant difference between the groups on psychiatric symptoms (i.e., depression, anxiety, and stress). IERITA emphasizes *behavioral* emotion regulation strategies: accepting internal experiences and controlling behaviors in emotional situations. Potentially, *cognitive* emotion regulation strategies (e.g., cognitive reappraisal, not included in IERITA) are specifically beneficial to reduce depressive and anxiety symptoms among youth;¹⁴⁷ how these findings apply to adolescents engaging in NSSI is something to investigate further. For adolescents with clinical levels of depression, anxiety, and stress, it is reasonable to include evidence-based interventions for these symptoms (e.g., behavioral activation, exposure, relaxation techniques) in parallel TAU. Consequently, it is crucial to evaluate if such TAU in combination with IERITA is advantageous for this subgroup.

If IERITA should be considered an efficacious treatment, the benefits should outweigh the adverse effects. A few (n = 4, 5%) participants in Study II reported adverse events related to IERITA at treatment termination. Reports included increased distress, sadness, and symptoms that the treatment addresses (self-destructive behaviors) and are likely to be reported by some receiving internet-delivered psychotherapy.¹²¹ Nevertheless, such adverse events are necessary to detect early to intervene, for instance, through enhancing therapist contact. A few individuals reported suicide attempts during the five-month-long study period (i.e., 5 [6%] in IERITA plus TAU and 9 [11%] in TAU-only). Based on these reports, there is no reason to believe that adding IERITA to TAU results in a higher number of suicide attempts, at least. As NSSI and suicidal behavior overlap,^{13,45,46,82-84} and only adolescents with immediate suicide risk were excluded, suicide attempts can be expected. However, based on previous findings,^{113,142,143} suicide attempts could have followed the same decreasing trend as NSSI in Study II. This hypothesis can be supported by the fact that NSSI, the strongest risk factor for suicide attempts,^{37,38} was reduced. Because suicide attempts are rarer than NSSI, larger samples are needed to investigate changes over time and between groups. In conclusion, adding IERITA to TAU can be more beneficial than harmful for adolescents with NSSI disorder on a group level.

4.3 Can emotion regulation be a mechanism of change in IERITA?

In Study II, mediation was investigated with a high resolution (weekly measures) of the theorized treatment mechanism and outcome. The results indicate that the effect of IERITA adjunctive to TAU, at least partly, goes through reducing emotion regulation difficulties. Hence, to optimize the effect of IERITA, the focus should be on decreasing

emotion regulation difficulties. There is still more to understand regarding *how* the change in emotion regulation difficulties the change NSSI frequency in IERITA (moving from mediator to mechanism). Results from Study I indicate that no specific skills taught in IERITA were superior in improving emotion regulation on a group level. Thus, which emotion regulation skill is most efficient for preventing NSSI can depend largely on individual and contextual factors.

There are still some crucial steps to establish emotion regulation as an empirically proven treatment mechanism. These steps include, for instance, investigating competing mediators; considering family functioning¹⁴⁸ or mentalization and attachment avoidance¹⁴⁹ as competing mediators are relevant as they have mediated outcomes in other treatments targeting adolescent self-injury. Investigating competing mediators can increase knowledge of shared or specific processes through which treatments for NSSI work. Secondly, demonstration of change in the mediator preceding change in the outcome and consistency across studies and samples are important. Emotion regulation difficulties have mediated treatment outcomes in other treatment studies targeting NSSI. It applies to adolescents receiving DBT-A,¹⁵⁰ and adults receiving Emotion Regulation Group Therapy (ERGT).¹⁵¹ In addition, the dose-response relationship found in Study II (the more reduction in emotion regulation difficulties, the more reduction in NSSI) strengthens the notion that emotion regulation can be a potential mechanism of change in IERITA.

4.4 What can we learn from the participant experience of IERITA?

The knowledge gained from Study I was beneficial for developing IERITA for Study II and has some implications worth discussing. First, investigating the supplementary app in a larger scale can increase the understanding of the added value of the app. Mobile apps can potentially be effective as a stand-alone intervention or in addition to face-to-face treatments.¹⁵² The mobile app was perceived as crucial for some adolescents. Adding mobile apps to complement face-to-face treatment can potentially increase the reach of digital aids to more extensive treatments. The flexibility in how to communicate with the therapist (online or telephone) or engage with the material (web-based material or mobile app) was appreciated in IERITA. Meeting treatment preferences can positively affect treatment adherence and outcome;¹⁵³ offering a toolbox of components to use in IERITA can thus hopefully facilitate optimal outcomes.

Our results from Study I indicate that therapist support is essential to both adolescents and parents. Also, many adolescents with other disorders have experienced that the online format can facilitate communication with the therapist.¹⁵⁴ If more or other formats (e.g., face-to-face/video meeting halfway through) of therapist contact could increase the efficacy of IERITA, should be considered further. Based on findings from Study I, how to combine skills and use them in challenging situations and how to structure treatment

work, are relevant topics for additional meetings. What therapist behaviors are most desirable should be investigated and can be informative in educating therapists and assessing therapist adherence.

Parents appreciated being involved in the treatment, and adolescents accepted their participation and could see the advantages of it. In addition, involving parents in a separate parent course was not associated with any apparent adverse effects. Nevertheless, some families may need additional family/parent interactive parts, as wished for by some parents in Study I. In fact, low family functioning can be a risk factor for NSSI.⁶⁵ Detecting and targeting low family functioning or high parental invalidation, as seen in Study III, is needed across conditions.

4.5 For whom can IERITA be more or less helpful?

IERITA was developed to create a treatment suitable for widespread use among youth with and without comorbidity. The results of Study III provided preliminary support for a broad utility as no strong or significant moderators were found. However, other variables that were not measured could still be of importance. It is also unknown if the results would differ if the outcome were operationalized differently, such as NSSI absence. Reductions on a group level are not likely to apply to every adolescent. A next relevant step can be investigating unaltered or decreasing NSSI trajectories and who is more likely to face such trajectories. Another possible future direction is to study the predictive effect of factors *during* treatment (e.g., credibility or initial response). Such evaluations can inform potential treatment modification (i.e., additional support) or decisions to switch to another treatment format. Larger samples are needed to investigate questions regarding differences in treatment response with greater certainty.

4.6 Can impulsivity and inattention in childhood be of importance to NSSI and suicidal behavior in adolescence?

The results from Study IV indicate that impulsivity and inattention are something to target across genders, possibly not only within ADHD. Impulsivity and inattention can be transient,¹⁶⁵ but still increase the risk of adverse outcomes at a later point. The findings from Study IV should be seen as one step of the way, where a central subsequent question is: Through what pathways do impulsivity and inattention increase the risk of NSSI and suicidal behavior? Longitudinal studies where exposure, mediator, and outcome are measured more than twice can help understand the associations in more detail and establish the temporal order. Potential mediators (factors to explain the association) to investigate further are, for instance, emotion dysregulation,¹⁵⁶ suicidal ideation,^{156,157} painful and proactive events,^{156,158}, and negative self-image and self-criticism⁷⁵. Such future studies are needed to develop effective early prevention strategies and understand the optimal time to intervene.

Given the connections between ADHD, emotion regulation, and self-injury,^{80,81,89-92} is it possible that targeting impulsivity and inattention in emotional situations is important to decrease the risk of self-injury? From the stance of the Experiential Avoidance Model,⁷⁴ youth with a high degree of impulsivity may need help to shift focus from short-term benefits to emotion regulation strategies with long-term benefits; youth with a high degree of inattention may need assistance to direct attention to select and retain situationally appropriate emotion regulation strategies. Emotion regulation treatment is promising not only for NSSI, but also in reducing impulsivity and inattention.⁸⁹ Thus, the hope is that IERITA, to some degree, could also reduce impulsivity and inattention symptoms. Some adolescents interested in IERITA will show elevated levels of impulsivity or inattention; for instance, in Study II, 18% had been diagnosed with ADHD.

4.7 General limitations

Designing a study of high quality is the aim of any researcher. Nonetheless, conducting research on potentially suicidal youth confers challenges that need to be handled with great care.¹³³ In Study II, the lack of balance between the groups (e.g., the group that received IERITA had additional material and an additional therapist contact in another format) limits internal validity concerning the specific effects of IERITA. In Study III, the statistical power was potentially constrained; however, it is common practice to base power calculations on the research question concerning efficacy on the primary outcome and evaluate moderators exploratively for novel treatments.¹⁵⁹ In Study IV, the sample size was pre-determined, and the estimated statistical power was low for effects of smaller magnitude. The expected effect is not always easy to determine when conducting power calculations, often because of limited previous research.

Some threats to the validity and reliability of measuring NSSI should be brought to attention. First, assessing NSSI can be challenging in some cases; for instance, separating NSSI from skin-picking and assessing the presence of tissue damage. In some cases, what defines one NSSI episode can be difficult to assess; if somebody stops self-injuring for a few minutes and then starts again, is this reported as one or two episodes? To overcome such challenges, interrater reliability evaluations and expert panel recommendations can be beneficial. Second, measurements of NSSI and suicidal behavior typically consist of self-reports, which entail certain risks of bias. The choice to disclose NSSI could be hindered by shame, fear of stigma and rejection.⁴³ However, when seeking treatment for NSSI (Studies I-III), willingness to disclose NSSI can increase.⁴³ Third, there is a risk of recall bias when reporting the occurrence of NSSI in retrospect. Hopefully, using weekly measures (Studies II-III) can decrease the risk of recall bias. Outcomes measured frequently can offer greater statistical power but should be balanced against the risk of missing values. In Studies II-III, the proportion of missing values was low for most weeks but relatively large for a few weeks. Missing values can threaten the validity of findings; thus, using efficient methods to deal with missing values

is essential. Likelihood-based models (used in Studies II-III) or multiple imputation are recommended methods under the assumption of missing at random.¹⁶⁰

The generalizability of the findings in Studies II–III can be threatened by low gender diversity (almost only girls). The underrepresentation of boys is a general limitation within this field of research¹⁰⁰ and regular care.⁵² In addition, NSSI is common among gender minorities⁵⁴ and thus desirable to have gender minorities represented in treatment studies. Including adolescents fulfilling NSSI disorder in Studies I–III entails a defined group of individuals (in comparison to having only a frequency criteria of NSSI). Nevertheless, NSSI disorder is not established, and the efficacy of IERITA adjunctive to TAU for subthreshold NSSI disorder remains unknown. Study IV included a defined but still heterogenic group: children at risk of neurodevelopmental disorders for different reasons. The generalizability is unclear as the study was underpowered to investigate differences based on reasons for being at risk of neurodevelopmental disorders.

5 Conclusions

The overarching aim of this thesis was to develop and evaluate Internet-delivered Emotion Regulation Individual Therapy for Adolescents (IERITA) with Nonsuicidal Self-Injury (NSSI) disorder and to improve our understanding of the early correlates of selfinjurious behaviors.

IERITA can be an acceptable, flexible, and credible treatment where therapist support is essential, and parents can successfully be involved. Nevertheless, additional support can be needed for those experiencing distress inherent to the online format. IERITA adjunctive to Treatment as Usual (TAU), is significantly more efficacious in reducing NSSI, emotion regulation difficulties, other self-destructive behaviors, and functional impairment compared to TAU only for adolescents with NSSI disorder. Additional trials are needed to understand the utility of IERITA across studies and samples. The mediation analysis strengthens the notion that emotion regulation can be a potential change mechanism in IERITA. For many, IERITA adjunctive to TAU can contribute to large, maintained reductions in NSSI with only a few or no NSSI episodes after treatment. Although the benefits of IERITA may outweigh the risks on a group level, adverse effects, including suicidal behavior, should be carefully monitored. IERITA can increase access to evidence-based NSSI treatment among a wide range of youth. These contributions could, by extension, innovate how we treat NSSI among adolescents.

For children at risk of neurodevelopmental disorders, detecting and treating impulsivity and inattention can be valuable to prevent self-injurious behaviors. Establishing the temporal precedence and investigating through potential pathways (e.g., emotion dysregulation) between impulsivity/inattention and NSSI/suicidal behavior are valuable further directions for optimal prevention and intervention.

6 Points of perspective

What does the future hold for the research field of adolescent Nonsuicidal Self-Injury (NSSI)? I hope the answer is "a lot" and I expect that we learn more in the following areas specifically during the upcoming years: (1) understanding the different presentations of NSSI (i.e., frequency, severity, persistence) in more detail and the impact of these; (2) how to validly categorize clinically significant NSSI; (3) the temporal associations between NSSI and biological, neurobiological, psychological, and social factors, not only through self-report; (4) treatment utility through well-designed treatment studies targeting NSSI (e.g., sufficient statistical power, balanced comparators, several clinically relevant outcomes measured, lengthy controlled follow-up). With that said, we will now focus on future directions and clinical implications of the studies in this thesis explicitly.

6.1 Replication

Replication studies are needed to become more confident of the actual effect of Internet-delivered Emotion Regulation Individual Therapy for Adolescents (IERITA). IERITA is currently being evaluated in a large-scale randomized clinical trial in Denmark (Clinical trial identifier: NCTO5179655). The strengths of this study include a large sample (300 adolescents) that will be recruited from within mental health services (clinical sample – increases external validity). Almost six years after starting Study II, other treatment studies involving internet-delivered treatment for adolescent NSSI are being conducted.¹⁶¹ This development creates opportunities to compare different kinds of treatments matched on the degree of therapist contact and treatment material in the future. Since designing Study II, investigator-delivered supportive therapy has been studied as a structured control condition for self-injury among adolescents.¹⁴² Supportive therapy can represent what many receive in regular health care services, as indicated by results from Study II. These progresses within the field are helpful in establishing the efficacy of treatments and can guide decisions regarding design questions in future studies.

6.2 Stepped care

Can IERITA be a potential first-line treatment for adolescents with NSSI disorder where more or other treatment can be initiated for those in need? If IERITA is cost-effective (as is currently being evaluated), more adolescents can be treated. Consequently, more resources can be put into families with more extensive treatment needs.

Testing IERITA in a stepped-care model for NSSI could be valuable to optimize outcomes. Getting inspiration from the stepped-care model in regular care for adult depression and anxiety in the United Kingdom, Improving Access to Psychological Therapies (IAPT),¹⁶² can be helpful; hence, a first step in a stepped-care model for NSSI

can entail a thorough assessment of symptom severity and functional impairment to decide on treatment intensity. For adolescent NSSI, adding a structured therapeutic assessment can be advantageous as it can increase subsequent treatment engagement and global functioning.^{100,163,164} During treatment, continuous monitoring and supervision are essential to guide decisions on moving to a more high-intensive alternative.¹⁶² What the more high-intensity alternatives for adolescent NSSI can comprise is an empirical question. Examples of high-intensity steps can include extending treatment length or increasing contact,^{103,104} switching to face-to-face or video sessions after IERITA (as has been beneficial for youth with obsessive-compulsive disorder¹⁶⁵), or other treatments, such as DBT-A - an extensive evidence-based treatment for self-injury.¹⁰³ Steppedcare models have, to my best knowledge, not been tested for NSSI among adolescents or adults. However, a stepped care model for adolescents at high risk of suicide is currently under investigation,¹⁶⁶ including internet-delivered treatment. A comprehensive treatment structure to treat NSSI among adolescents could benefit patients (e.g., getting the best possible treatment based on current needs) and organizations (e.g., guidance for clinicians on what steps to take if insufficient treatment effect is needed).

6.3 Implementation

The ambition is that IERITA, given evidence of its efficacy, can constitute a low-threshold treatment for a wide range of adolescents with NSSI. To find adolescents that can benefit from IERITA, health care professionals need to screen for NSSI routinely. In addition, health care professionals need to be educated about NSSI and how to respond in helpful, non-stigmatizing ways.⁴³ One of the main benefits of internet-delivered treatment is scalability. What key steps must be taken to achieve this for IERITA? First, IERITA must be implemented on a platform that can be used within health care services, preferably nationally. To successfully implement IERITA, inspiration from the experiences of the IAPT program¹⁶² can be helpful; for instance, creating a platform with sufficiently trained IERITA therapists, central administration, and a high degree of regular supervision. In addition, safety, assessment, and evaluation procedures used in the research studies need to be applied to suit the health care structure.

Evaluating the implementation of IERITA should be prioritized, given that there is only one published large-scale randomized clinical trial. Several testable questions may arise in the implementation process, including but not limited to: Who is it reasonable to offer IERITA to – could the eligibility from the studies be modified? How many of those offered IERITA accept and adhere to treatment? What are the reasons for nonparticipation or low adherence? How does IERITA work in combination with different types of TAU? How can IERITA be optimized to improve treatment outcomes? What is the therapists' experience from working with IERITA – what are the challenges?

6.4 Prevention

Screening children and adolescents for impulsivity and inattention, regardless of primary reason for contact, is easily done within mental health services. If high levels of impulsivity or inattention are detected, offering interventions targeting impulse control and attentional ability can be valuable. In addition, children, or adolescents with high levels of impulsivity and inattention should be monitored concerning self-injury and associated correlates (e.g., suicidal ideation, emotion dysregulation).

7 Acknowledgements

First, I want to express my appreciation to all the children, adolescents, and parents participating in these studies. Your contribution makes the field move forward and help other families in the future. Thank you!

Clara Hellner, my main supervisor. Thank you for believing in my potential and for being so supportive and encouraging throughout all these years! I am lucky to have had you as a supervisor – you are a star! Your endless curiosity, sharpness, and ability to get things done are apparent in every room you enter. I am amazed and inspired by your abilities as a leader, researcher, clinician, speaker, and writer. Coordinating a large-scale treatment study is not always easy and I appreciate all the phone calls we have had (sometimes short yet important). I am looking forward to continuing the IERITA journey together!

Johan Bjureberg, my co-supervisor. I cannot thank you enough for creating this opportunity for me and wholeheartedly sharing your top-notch clinical and research skills. You are so generous with your time and always with a smile on your face. Few are as talented, clever, dedicated, and caring as you. I don't understand how you can do everything you do, but being around you makes one believe that dreams do come true (as cheesy as it may sound). You are already a legend within the field of adolescent NSSI, suicidal behavior, and emotion regulation, and a big inspiration for me! I am thrilled about our continued collaboration!

Ralf Kuja–Halkola, my co-supervisor. King of statistics! Thank you for taking the time and so patiently teaching me statistics. You are one of the most pedagogical persons I know, and impressively kind and brainy! After discussions with you, I understand everything I don't understand, but that is precisely what I need. Being around you makes me eager to learn more, and I feel even more confident now that statistical knowledge is power!

Sebastian Lundström, my mentor. I have really appreciated our meetings and that you always value the clinical perspective. You are a role model when it comes to combining research with clinical practice. Thank you for introducing me to the research community in Gothenburg and sending loads of positive reinforcements after my half-time control!

The rest of the Bjureberg/Hellner research group, *Anna Ohlis* (Thanks for leading the way! It has been a pleasure following your footsteps!), *Jannike Nilbrink, Katja Sjöblom, Lars Klintwall, Martin Bellander, Moa Pontén, Moa Karemyr* and *Oskar Flygare*. Thank you for including me in your warm, fun, and passionate environment. I never work as good as after hanging out with you guys. Our paper-in-a-day sessions have been epic! The food, musical performances, giggles! Honestly, the competence, kindness, and humor this group possesses is astonishing. Remember these names; I am sure great things will come for all of you.

Co-authors *Brjánn Ljótsson* (Thanks for being the main supervisor to our master thesis and for hosting the BASS platform!) *Emma Claesdotter–Knutsson, Erik Hedman–Lagerlöf, Eva Norén Selinus, Hanna Sahlin, Hedvig Engberg* (Thanks for teaching me qualitative methods so pedagogically and eagerly!), *Henrike Häbel, Henrik Larsson, Hugo Hesser* (Thanks for sharing your impressive statistical knowledge so generously!), *Julia Stensils, Kim L Gratz, Martin Cederlöf, Matthew T Tull, Paul Lichtenstein,* and *Sebastian Lundström.* My biggest thank you for the fantastic collaborations on the included studies and for your invaluable comments on manuscripts. I am happy to have learned the true power of teamwork!

Maria Åbonde Garke, my desk partner in crime for many years. Thank you for chairing my dissertation and being a big sister – PhD student style. Thanks for always calmly answering my million questions and all the singing (I miss it!).

Colleagues at Centre for Psychiatry Research. Thank you, *Nitya Jayaram–Lindström*, *Lena Melander*, and the rest of the stab, for your dedication to keeping the workspace inspiring and making use of all the unique competence there is. Thanks to the CPF COVID project Alexander Rozental, David Forsström, Johan Lundin, Karolina Sörman, *Kristoffer N T Månsson, Maria Hedman–Lagerlöf, Nitya Jayaram–Lindström, Samir El* Alaoui, Shervin Shahnavaz, Simon Jangard, and Tobias Lundgren for all the fruitful discussions and fun late–night sessions. It has been incredible to experience the power of getting together to act in times of societal crisis. Thank you, Anders Nilsson, for an enjoyable collaboration and the opportunity to deepen my knowledge of qualitative methods. Thanks for all the helpful discussions in journal clubs over the years. Especially thanks to Kristoffer Magnusson, Karolina Sörman, Maria Åbonde Garke, and Philip Lindner for making these sessions possible.

Collegues at Barninternetprojektet (BIP). Sarah Vigerland and Eva Serlachius, thank you for leading the way and generously sharing your experience and passion to increase access to evidence-based psychosocial treatments for children and adolescents. *Rebecca Grudin*, thank you for all the support, discussions on study coordination and life, and laughs during our PhD journeys. It has been so much fun lecturing together about adverse events and harms. I am sure you have a brilliant career ahead of you, and I hope we can continue collaborating.

All clinicians in IERITA. Alexandra Boalt, Ellen Isaksson, Erzsébet Holmquist, Fredrika Zeidlitz, Jenny Lundgren, Jon Hedlund, Katja Sjöblom, Katri Frankenstein, Malin Olsen, Maria Holmgren, Marie Werner, Niclas Borg, Rebecca Grudin, Sofia Dinnetz, and Vera Wachtmeister. Without you, it would not have been possible to conduct the RCT. Thank you for your endless patience with routines, documentation, and hard work supporting the families over the years. *Rébecca Ankarvall Marianne,* thank you for assisting us with administrative work in the RCT. You were greatly appreciated!

Colleagues at BUP OCD and related disorders. *Maria Silverberg Mörse*, thank you for taking me on as a psychologist during training. *Linn Lichtenstein* and *Maria Arendt Hillborg*, thank you for supervising me and generously sharing your incredible clinical expertise. Thank you all for forcing me to take a coffee break every now and then, for all the delicious Thursday breakfasts, and for a clinically inspiring environment – wow, you are excellent clinicians, all of you!

National Self-harm Project and the patient organization SHEDO. *Eva Klingberg,* thank you for our many fun and uplifting chats by the desks. I miss working together!

Collaborators in Denmark who are working with the TEENS trial. Thank you for your positive vibes, fun dinners, essential questions, and input. Look forward to continuing the collaboration!

I would also like to thank all the funders for making these studies possible: National Selfharm Project, the Swedish Research Council, the Stockholm County Council, the ALF medicine projects in Sweden, Markus and Amelia Wallenberg Foundation, Fredrik and Ingrid Thuring's Foundation, Clas Groschinsky's Foundation, Sven Jerring Foundation, Kempe-Carlgrenska Foundation, and Bror Gadelius Foundation.

Psychologist friends from education, Kosegruppa, *Evelina, Julia, Nathalie, Madde*, and *Sarah*, thanks for all the study group sessions, travels, and parties. I miss hanging out!

Friends from childhood and school. *Elin, Ellen, Frida x2, Josefin, Katta, Liv, Malin,* and *Petter*. Thanks for being such amazing friends that I can enjoy life with.

My family, for support, love, and boost throughout these years and life. *Mamma Lena* and *pappa Stig*, you are the best kind of cheerleaders, never with pressure and always with belief. *Clara*, my baby (taller and smarter) sister. Thanks for letting me talk about R and STATA for far too long and always being there <3. *Tor*, thanks for being such a caring boyfriend to Clara and letting us stay at your place when we visit Stockholm. *Mormor Birgit. Tack för fina samtal och varma kramar.* Family in–law, thanks for inspiring me to an academic career and for making me feel so included since day one. Also, thanks for believing in me to learn tennis, golf, and padel and never giving up on me despite my lack of talent. I am so lucky to have all of you in my life!

My husband *Nicholas*, you are the best (and you know it!), and I am so lucky to have had your love and never-ending support for the past 13 years. Thank you for moving to Stockholm (forcing me to move) and encouraging me to study psychology and research. During this thesis work, we have managed to build a house, move, postpone our wedding five times, finally get married, and have a baby. It has been some busy, challenging, and exciting years for sure, but nothing we can't manage. *Jag älskar dig*.

My daughter Ellie. Mamma älskar dig, alltid.

8 References

- Angelotta C. Defining and Refining Self-Harm: A Historical Perspective on Nonsuicidal Self-Injury. *The journal of nervous and mental disease*. 2015; 203(2): 75–80.
- 2. Muehlenkamp JJ. Self-Injurious Behavior as a Separate Clinical Syndrome. American Journal of Orthopsychiatry. 2005; 75(2): 324–33.
- International Society for the Study of Self-injury. What is self-injury? 2023. <u>https://www.itriples.org/what-is-nssi</u> (accessed 2023 09 26).
- Zetterqvist M. The DSM-5 diagnosis of nonsuicidal self-injury disorder: a review of the empirical literature. *Child and Adolescent Psychiatry and Mental Health*. 2015; 9(1): 31-.
- 5. Nock M. Self-injury. Annual Review of Clinical Psychology. 2010; 6: 339-63.
- Kapur N, Cooper J, O'Connor RC, Hawton K. Non- suicidal self- injury v. attempted suicide: new diagnosis or false dichotomy? *The British Journal of Psychiatry*. 2013; 202(5): 326.
- 7. American Psychiatric Association. *Diagnostic and statistical manual of mental disorders*. 5 ed. Arlington, VA: American Psychiatric Publishing; 2013.
- 8. St. Germain SA, Hooley JM. Direct and indirect forms of non-suicidal self-injury: Evidence for a distinction. *Psychiatry Research*. 2012; 197(1-2): 78-84.
- Jonsson LS, Svedin CG, Priebe G, Fredlund C, Wadsby M, Zetterqvist M. Similarities and Differences in the Functions of Nonsuicidal Self-Injury (NSSI) and Sex as Self-Injury (SASI). Suicide & life-threatening behavior. 2019; 49(1): 120–36.
- Hooley JM, Fox KR, Boccagno C. Nonsuicidal Self-Injury: Diagnostic Challenges And Current Perspectives. *Neuropsychiatric Disease and Treatment*. 2020; 16: 101.
- Andrewes HE, Hulbert C, Cotton SM, Betts J, Chanen AM. Relationships between the frequency and severity of non-suicidal self-injury and suicide attempts in youth with borderline personality disorder. *Early Intervention in Psychiatry*. 2017; 13(2): 194–201.
- Sleuwaegen E, Houben M, Claes L, Berens A, Sabbe B. The relationship between non-suicidal self-injury and alexithymia in borderline personality disorder: "Actions instead of words". *Comprehensive Psychiatry*. 2017; 77: 80–8.
- Nock MK, Joiner TE, Jr, Gordon KH, Lloyd-Richardson E, Prinstein MJ. Non-suicidal self-injury among adolescents: diagnostic correlates and relation to suicide attempts. *Psychiatry Research*. 2006; 144(1): 65–72.

- Glenn C, Klonsky D. Nonsuicidal Self-Injury Disorder: An Empirical Investigation in Adolescent Psychiatric Patients. *Journal of Clinical Child & Adolescent Psychology*. 2013; 42(4): 496–507.
- Buelens T, Costantini G, Luyckx K, Claes L. Comorbidity Between Non-suicidal Self-Injury Disorder and Borderline Personality Disorder in Adolescents: A Graphical Network Approach. *Frontiers in psychiatry*. 2020; 11: 580922–.
- In-Albon T, Ruf C, Schmid M. Proposed Diagnostic Criteria for the of Nonsuicidal Self-Injury in Female Adolescents: Diagnostic and Clinical Correlates. *Psychiatry Journal*. 2013; 2013: 159208.
- Wilkinson P. Non-suicidal self-injury. European Child & Adolescent Psychiatry. 2013; 20(2): 103–8.
- 18. Pattison EM, Kahan J. The deliberate self-harm syndrome. *The American journal* of psychiatry. 1983; 140(7): 867–72.
- 19. Favazza AR, Rosenthal RJ. Diagnostic Issues in Self-Mutilation. *Psychiatric services (Washington, DC)*. 1993; 44(2): 134–40.
- 20. Brausch A. *Diagnostic Classification of Nonsuicidal Self-Injury*. In: Washburn JJ, ed. Nonsuicidal Self-injury: Advances in Research and Practice: Routledge; 2019.
- Washburn JJ, Potthoff LM, Juzwin KR, Styer DM. Assessing DSM-5 Nonsuicidal Self-Injury Disorder in a Clinical Sample. *Psychological Assessment*. 2015; 27(1): 31–41.
- 22. Andover MS. Non-suicidal self-injury disorder in a community sample of adults. *Psychiatry Research*. 2014; 219(2): 305-10.
- 23. Muehlenkamp JJ, Brausch AM, Washburn JJ. How Much Is Enough? Examining Frequency Criteria for NSSI Disorder in Adolescent Inpatients. *Journal of Consulting and Clinical Psychology*. 2017; 85(6): 611–9.
- Selby EA, Kranzler A, Fehling KB, Panza E. Nonsuicidal self-injury disorder: The path to diagnostic validity and final obstacles. *Clinical Psychology Review*. 2015; 38: 79–91.
- 25. Plener PL, Schumacher TS, Munz LM, Groschwitz RC. The longitudinal course of non-suicidal self-injury and deliberate self-harm: a systematic review of the literature. *Borderline Personality Disorder and Emotion Dysregulation*. 2015; 2: 2.
- Gandhi A, Luyckx K, Baetens I, et al. Age of onset of non-suicidal self-injury in Dutch-speaking adolescents and emerging adults: An event history analysis of pooled data. *Comprehensive Psychiatry*. 2018; 80: 170–8.
- Glenn CR, Lanzillo EC, Esposito EC, Santee AC, Nock MK, Auerbach RP. Examining the Course of Suicidal and Nonsuicidal Self-Injurious Thoughts and Behaviors in Outpatient and Inpatient Adolescents. *Journal of abnormal child psychology*. 2016; 45(5): 971-83.

- Mars B, Heron J, Crane C, et al. Clinical and social outcomes of adolescent self harm: population based birth cohort study. *BMJ*. 2014; 349: g5954.
- Moran P, Coffey C, Romaniuk H, et al. The natural history of self-harm from adolescence to young adulthood: a population-based cohort study. *Lancet*. 2012; 379(9812): 236–43.
- 30. Turner BJ, Helps CE, Ames ME. Stop self-injuring, then what? Psychosocial risk associated with initiation and cessation of nonsuicidal self-injury from adolescence to early adulthood. *Journal of abnormal psychology* (1965). 2021.
- Duggan J, Heath N, Hu T. Non-suicidal self-injury maintenance and cessation among adolescents: a one-year longitudinal investigation of the role of objectified body consciousness, depression and emotion dysregulation. *Child* and adolescent psychiatry and mental health. 2015; 9(1): 21–.
- 32. Kiekens G, Hasking P, Bruffaerts R, et al. What Predicts Ongoing Nonsuicidal Self-Injury?: A Comparison Between Persistent and Ceased Self-Injury in Emerging Adults. *The journal of nervous and mental disease*. 2017; 205(10): 762-70.
- 33. Kelada L, Hasking P, Melvin G, Whitlock J, Baetens I. "I Do Want to Stop, At Least I Think I Do": An International Comparison of Recovery From Nonsuicidal Self-Injury Among Young People. *Journal of adolescent research*. 2018; 33(4): 416–41.
- Gelinas BL, Wright KD. The Cessation of Deliberate Self-Harm in a University Sample: The Reasons, Barriers, and Strategies Involved. Archives of Suicide Research. 2013; 17(4): 373–86.
- Wilkinson PO, Qiu T, Neufeld S, Jones PB, Goodyer IM. Sporadic and recurrent non-suicidal self-injury before age 14 and incident onset of psychiatric disorders by 17 years: prospective cohort study. *The British Journal of Psychiatry*. 2018; 212(4): 222.
- 36. Bjureberg J, Ohlis A, Ljótsson B, et al. Adolescent self-harm with and without suicidality: cross-sectional and longitudinal analyses of a Swedish regional register. *Journal of Child Psychology and Psychiatry*. 2019; 60(3): 295-304.
- Ribeiro JD, Franklin JC, Fox KR, et al. Self-injurious thoughts and behaviors as risk factors for future suicide ideation, attempts, and death: a meta-analysis of longitudinal studies. *Psychological Medicine*. 2016; 46(2): 225–36.
- Castellví P, Lucas-Romero E, Miranda-Mendizábal A, et al. Longitudinal association between self-injurious thoughts and behaviors and suicidal behavior in adolescents and young adults: A systematic review with meta-analysis. *Journal* of affective disorders. 2017; 215: 37–48.
- World Health Organization. Global health estimates: Leading causes of death.
 2020. <u>https://www.who.int/data/gho/data/themes/mortality-and-global-health-estimates/ghe-leading-causes-of-death</u> (accessed 2023 09 24).

- Koenig J, Brunner R, Fischer-Waldschmidt G, et al. Prospective risk for suicidal thoughts and behaviour in adolescents with onset, maintenance or cessation of direct self-injurious behaviour. *European Child & Adolescent Psychiatry*. 2017; 26(3): 345–54.
- 41. Swannell SV, Martin GE, Page A, Hasking P, St John NJ. Prevalence of Nonsuicidal Self- Injury in Nonclinical Samples: Systematic Review, Meta-Analysis and Meta-Regression. *Suicide and Life-Threatening Behavior*. 2014; 44(3): 273–303.
- 42. Muehlenkamp J, Claes L, Havertape L, Plener Paul L. International prevalence of adolescent non-suicidal self-injury and deliberate self-harm. *Child and Adolescent Psychiatry and Mental Health.* 2012; 6(1): 10.
- 43. Simone AC, Hamza CA. Examining the disclosure of nonsuicidal self-injury to informal and formal sources: A review of the literature. *Clinical psychology review*. 2020; 82: 101907–.
- Gillies D, Christou MA, Dixon AC, et al. Prevalence and Characteristics of Self– Harm in Adolescents: Meta–Analyses of Community–Based Studies 1990–2015. Journal of the American Academy of Child and Adolescent Psychiatry. 2018; 57(10): 733–41.
- 45. Asarnow JR, Porta G, Spirito A, et al. Suicide Attempts and Nonsuicidal Self-Injury in the Treatment of Resistant Depression in Adolescents: Findings from the TORDIA Study. *Journal of the American Academy of Child & Adolescent Psychiatry*. 2011; 50(8): 772–81.
- Jacobson CM, Muehlenkamp JJ, Miller AL, Turner JB. Psychiatric Impairment Among Adolescents Engaging in Different Types of Deliberate Self- Harm. *Journal* of Clinical Child & Adolescent Psychology. 2008; 37(2): 363–75.
- 47. Kaess M, Parzer P, Mattern M, et al. Adverse childhood experiences and their impact on frequency, severity, and the individual function of nonsuicidal self-injury in youth. *Psychiatry Research*. 2013; 206(2-3): 265–72.
- Zetterqvist M, Lundh LG, Dahlstrom O, Svedin CG. Prevalence and function of non-suicidal self-injury (NSSI) in a community sample of adolescents, using suggested DSM-5 criteria for a potential NSSI disorder. *Journal of Abnormal Child Psychology*. 2013; 41(5): 759–73.
- 49. Buelens T, Luyckx K, Kiekens G, Gandhi A, Muehlenkamp JJ, Claes L. Investigating the DSM-5 criteria for non-suicidal self-injury disorder in a community sample of adolescents. *Journal of affective disorders*. 2020; 260: 314–22.
- 50. Kiekens G, Hasking P, Claes L, et al. The DSM-5 nonsuicidal self-injury disorder among incoming college students: Prevalence and associations with 12-month mental disorders and suicidal thoughts and behaviors. *Depression and anxiety*. 2018; 35(7): 629–37.

- Polanczyk GV, Salum GA, Sugaya LS, Caye A, Rohde LA. Annual Research Review: A meta-analysis of the worldwide prevalence of mental disorders in children and adolescents. *Journal of child psychology and psychiatry*. 2015; 56(3): 345–65.
- 52. Bresin K, Schoenleber M. Gender differences in the prevalence of nonsuicidal self-injury: A meta-analysis. *Clinical Psychology Review*. 2015; 38: 55–64.
- Kimbrel NA, Calhoun PS, Beckham JC. Nonsuicidal self-injury in men: a serious problem that has been overlooked for too long. *World Psychiatry*. 2017; 16(1): 108– 9.
- 54. Liu RT, Sheehan AE, Walsh RFL, Sanzari CM, Cheek SM, Hernandez EM. Prevalence and correlates of non-suicidal self-injury among lesbian, gay, bisexual, and transgender individuals: A systematic review and meta-analysis. *Clinical Psychology Review*. 2019; 74: 101783.
- 55. Jackman K, Honig J, Bockting W. Nonsuicidal self-injury among lesbian, gay, bisexual and transgender populations: an integrative review. *Journal of Clinical Nursing*. 2016; 25(23–24): 3438–53.
- McManus S, Gunnell D, Cooper C, et al. Prevalence of non-suicidal self-harm and service contact in England, 2000–14: repeated cross-sectional surveys of the general population. *The Lancet Psychiatry*. 2019; 6(7).
- 57. Wester K, Trepal H, King K. Nonsuicidal Self-Injury: Increased Prevalence in Engagement. *Suicide and Life-Threatening Behavior*. 2018; 48(6): 690–8.
- Morgan C, Webb RT, Carr MJ, et al. Incidence, clinical management, and mortality risk following self harm among children and adolescents: cohort study in primary care. *BMJ*. 2017; 359: j4351.
- Zetterqvist M, Jonsson LS, Landberg Å, Svedin CG. A potential increase in adolescent nonsuicidal self-injury during covid-19: A comparison of data from three different time points during 2011 – 2021. *Psychiatry research*. 2021; 305: 114208-.
- Jarvi S, Jackson B, Swenson L, Crawford H. The Impact of Social Contagion on Non-Suicidal Self-Injury: A Review of the Literature. *Archives of Suicide Research*. 2013; 17(1): 1–19.
- Brown RC, Fischer T, Goldwich AD, Keller F, Young R, Plener PL. cutting: Nonsuicidal self-injury (NSSI) on Instagram. *Psychological medicine*. 2018; 48(2): 337– 46.
- Gunnell D, Kidger J, Elvidge H. Adolescent mental health in crisis. *BMJ*. 2018; 361: k2608-k.
- Franklin JC, Ribeiro JD, Fox KR, et al. Risk Factors for Suicidal Thoughts and Behaviors: A Meta-Analysis of 50 Years of Research. *Psychological Bulletin*. 2017; 143(2): 187–232.

- Nock M, K. Why Do People Hurt Themselves? New Insights Into the Nature and Functions of Self-Injury. *Current Directions in Psychological Science*. 2009; 18(2): 78-83.
- Fox KR, Franklin JC, Ribeiro JD, Kleiman EM, Bentley KH, Nock MK. Meta-analysis of risk factors for nonsuicidal self-injury. *Clinical Psychology Review*. 2015; 42: 156– 67.
- Wang Y-J, Li X, Ng CH, Xu D-W, Hu S, Yuan T-F. Risk factors for non-suicidal selfinjury (NSSI) in adolescents: A meta-analysis. *EClinicalMedicine*. 2022; 46: 101350-.
- Maciejewski DF, Creemers HE, Lynskey MT, et al. Overlapping Genetic and Environmental Influences on Nonsuicidal Self-injury and Suicidal Ideation: Different Outcomes, Same Etiology? JAMA Psychiatry. 2014; 71(6): 699–705.
- Kaess M, Hooley JM, Klimes-Dougan B, et al. Advancing a temporal framework for understanding the biology of nonsuicidal self- injury: An expert review. *Neuroscience and biobehavioral reviews*. 2021; 130: 228–39.
- 69. Caicedo S, Whitlock J. *Top misconceptions about self-injury*. Ithaca, NY: Cornell University; 2009.
- Klonsky ED, Glenn CR, Styer DM, Olino TM, Washburn JJ. The functions of nonsuicidal self-injury: converging evidence for a two-factor structure. *Child and Adolescent Psychiatry and Mental Health.* 2015; 9(1): 44–.
- 71. Taylor PJ, Jomar K, Dhingra K, Forrester R, Shahmalak U, Dickson JM. A metaanalysis of the prevalence of different functions of non-suicidal self-injury. *Journal of Affective Disorders*. 2018; 227: 759–69.
- Victor SE, Styer D, Washburn JJ. Functions of nonsuicidal self-injury (NSSI): Cross-sectional associations with NSSI duration and longitudinal changes over time and following treatment. *Psychiatry Research*. 2016; 241: 83–90.
- Nock MK. Actions speak louder than words: An elaborated theoretical model of the social functions of self-injury and other harmful behaviors. *Applied & preventive psychology*. 2008; 12(4): 159–68.
- Chapman AL, Gratz KL, Brown MZ. Solving the puzzle of deliberate self-harm: The experiential avoidance model. *Behaviour Research and Therapy*. 2006; 44(3): 371–94.
- Hooley JM, Franklin JC. Why Do People Hurt Themselves? A New Conceptual Model of Nonsuicidal Self-Injury. *Clinical Psychological Science*. 2018; 6(3): 428– 51.
- 76. Thompson RA. Emotion regulation: a theme in search of definition. *Monographs of the Society for Research in Child Development*. 1994; 59(2-3): 25–52.

- 77. Gratz KL, Roemer L. Multidimensional Assessment of Emotion Regulation and Dysregulation: Development, Factor Structure, and Initial Validation of the Difficulties in Emotion Regulation Scale. *Journal of Psychopathology and Behavioral Assessment*. 2004; 26(1): 41–54.
- Aldao A, Nolen-Hoeksema S, Schweizer S. Emotion-regulation strategies across psychopathology: A meta-analytic review. *Clinical Psychology Review*. 2010; 30(2): 217–37.
- Compas BE, Jaser SS, Bettis AH, et al. Coping, Emotion Regulation, and Psychopathology in Childhood and Adolescence: A Meta-Analysis and Narrative Review. *Psychological Bulletin*. 2017; 143(9): 939–91.
- Hemming L, Taylor P, Haddock G, Shaw J, Pratt D. A systematic review and metaanalysis of the association between alexithymia and suicide ideation and behaviour. *Journal of affective disorders*. 2019; 254: 34–48.
- 81. Neacsiu AD, Fang CM, Rodriguez M, Rosenthal MZ. Suicidal Behavior and Problems with Emotion Regulation. *Suicide & life-threatening behavior*. 2018; 48(1): 52–74.
- Brausch AM, Gutierrez PM. Differences in Non-Suicidal Self-Injury and Suicide Attempts in Adolescents. *Journal of youth and adolescence*. 2009; 39(3): 233– 42.
- Muehlenkamp JJ, Gutierrez PM. Risk for Suicide Attempts Among Adolescents Who Engage in Non-Suicidal Self-Injury. *Archives of Suicide Research*. 2007; 11(1): 69–82.
- Ghinea D, Edinger A, Parzer P, Koenig J, Resch F, Kaess M. Non-suicidal self-injury disorder as a stand-alone diagnosis in a consecutive help-seeking sample of adolescents. *Journal of affective disorders*. 2020; 274: 1122–5.
- Zareian B, Klonsky DE. Nonsuicidal and Suicidal Self-injury. In: Washburn JJ, ed. Nonsuicidal Self-Injury: Advances in Research and Practice. New York: Routledge; 2019.
- Hamza CA, Stewart SL, Willoughby T. Examining the link between nonsuicidal selfinjury and suicidal behavior: A review of the literature and an integrated model. *Clinical Psychology Review*. 2012; 32(6): 482–95.
- Joiner T. Why people die by suicide. Cambridge, MA: Harvard University Press; 2005.
- Bosch R, Pagerols M, Rivas C, et al. Neurodevelopmental disorders among Spanish school-age children: prevalence and sociodemographic correlates. *Psychological medicine*. 2021: 1–11.
- Shaw P, Stringaris A, Nigg J, Leibenluft E. Emotion Dysregulation in Attention Deficit Hyperactivity Disorder. *American Journal of Psychiatry*. 2014; 171(3): 276– 93.

- 90. Meszaros G, Horvath LO, Balazs J. Self-injury and externalizing pathology: a systematic literature review. *BMC psychiatry*. 2017; 17(1): 160.
- 91. Allely CS. The association of ADHD symptoms to self-harm behaviours: a systematic PRISMA review. *BMC Psychiatry*. 2014; 14(1): 133.
- Septier M, Stordeur C, Zhang J, Delorme R, Cortese S. Association between suicidal spectrum behaviors and Attention-Deficit/Hyperactivity Disorder: A systematic review and meta-analysis. *Neuroscience and biobehavioral reviews*. 2019; 103: 109–18.
- 93. Anckarsäter H, Larson T, Hansson SL, et al. Child neurodevelopmental and behavioural problems are intercorrelated and dimensionally distributed in the general population. *The Open Psychiatry Journal*. 2008; 2(1).
- 94. Lubke GHPD, Hudziak JJMD, Derks EMPD, van Bijsterveldt TCEMPD, Boomsma DIPD. Maternal Ratings of Attention Problems in ADHD: Evidence for the Existence of a Continuum. *Journal of the American Academy of Child and Adolescent Psychiatry*. 2009; 48(11): 1085–93.
- 95. Gerrard LD. Attention Deficit Hyperactivity Disorder Symptoms and Nonsuicidal Self-Injury Among Clinically-Referred Children and Youth. ProQuest Dissertations Publishing: University of Toronto, Canada; 2018.
- 96. Doyle RL, Fite PJ. Indicators of suicidal outcomes among 6- to 12-year-old treatment seeking youth. *Child Psychiatry & Human Development*. 2021.
- 97. Hamza CA, Willoughby T, Heffer T. Impulsivity and nonsuicidal self-injury: A review and meta-analysis. *Clinical Psychology Review*. 2015; 38: 13–24.
- Liu RT, Trout ZM, Hernandez EM, Cheek SM, Gerlus N. A behavioral and cognitive neuroscience perspective on impulsivity, suicide, and non-suicidal self-injury: Meta-analysis and recommendations for future research. *Neuroscience and biobehavioral reviews*. 2017; 83: 440–50.
- 99. Meza JI, Owens EB, Hinshaw SP. Childhood predictors and moderators of lifetime risk of self-harm in girls with and without attention-deficit/hyperactivity disorder. *Development and psychopathology.* 2020; 33(4): 1351–67.
- Witt KG, Hawton K, Hetrick SE, Taylor Salisbury TL, Townsend E, Hazell P. Interventions for self-harm in children and adolescents. *Cochrane database of systematic reviews*. 2020; 2020(7).
- 101. Iyengar U, Snowden N, Asarnow JR, Moran P, Tranah T, Ougrin D. A Further Look at Therapeutic Interventions for Suicide Attempts and Self-Harm in Adolescents: An Updated Systematic Review of Randomized Controlled Trials. *Frontiers in Psychiatry*. 2018; 9.
- 102. Flaherty H. Treating Adolescent Nonsuicidal Self-Injury: A Review of Psychosocial Interventions to Guide Clinical Practice. *Child and Adolescent Social Work Journal*. 2018; 35(1): 85–95.
- 103. Glenn CR, Esposito EC, Porter AC, Robinson DJ. Evidence Base Update of Psychosocial Treatments for Self-Injurious Thoughts and Behaviors in Youth. *Journal of Clinical Child and Adolescent Psychology*. 2019; 48(3): 357.
- 104. Glenn CR, Franklin JC, Nock MK. Evidence-based psychosocial treatments for self-injurious thoughts and behaviors in youth. *Journal of Clinical Child* & Adolescent Psychology. 2015; 44(1): 1-29.
- 105. Kothgassner OD, Robinson K, Goreis A, Ougrin D, Plener PL. Does treatment method matter? A meta-analysis of the past 20 years of research on therapeutic interventions for self-harm and suicidal ideation in adolescents. *Borderline Personality Disorder and Emotion Dysregulation*. 2020; 7: 9.
- 106. Ougrin D, Tranah T, Stahl D, Moran P, Asarnow JR. Therapeutic interventions for suicide attempts and self-harm in adolescents: systematic review and metaanalysis. *Journal of the American Academy of Child & Adolescent Psychiatry*. 2015; 54(2): 97–107 e2.
- Harris LM, Huang X, Funsch KM, Fox KR, Ribeiro JD. Efficacy of interventions for suicide and self-injury in children and adolescents: a meta-analysis. *Scientific* reports. 2022; 12(1): 12313–.
- Turner BJ, Austin SB, Chapman AL. Treating nonsuicidal self-injury: a systematic review of psychological and pharmacological interventions. *Canadian Journal of Psychiatry*. 2014; 59(11): 576–85.
- Nationellt system för kunskapsstyrning. Nationellt vård- och insatsprogram för självskadebeteende. 2023. <u>https://www.vardochinsats.se/sjaelvskadebeteende/</u> (accessed 2023 09 15).
- 110. National Institute for Health and Care Excellence. Self-harm: assessment, management and preventing recurrence. 2022. <u>https://www.nice.org.uk/guidance/ng225/chapter/Recommendations#interventions=for-self-harm</u> (accessed 2023 09 21).
- 111. Wilkinson P, Kelvin R, Roberts C, Dubicka B, Goodyer I. Clinical and Psychosocial Predictors of Suicide Attempts and Nonsuicidal Self-Injury in the Adolescent Depression Antidepressants and Psychotherapy Trial (ADAPT). American Journal of Psychiatry. 2011; 168(5): 495–501.
- Asarnow JR, Ougrin D. Editorial: Suicide and self-harm: advancing from science to preventing deaths. *Journal of Child Psychology and Psychiatry*. 2019; 60(10): 1043–5.
- 113. Kaess M, Edinger A, Fischer-Waldschmidt G, Parzer P, Brunner R, Resch F. Effectiveness of a brief psychotherapeutic intervention compared with

treatment as usual for adolescent nonsuicidal self-injury: a single-centre, randomised controlled trial. *European Child & Adolescent Psychiatry*. 2019; 29(6): 881–91.

- 114. Gratz KL, Gunderson JG. Preliminary Data on an Acceptance-Based Emotion Regulation Group Intervention for Deliberate Self-Harm Among Women With Borderline Personality Disorder. *Behavior therapy*. 2006; 37(1): 25–35.
- 115. Gratz KL, Tull MT, Levy R. Randomized controlled trial and uncontrolled 9-month follow-up of an adjunctive emotion regulation group therapy for deliberate selfharm among women with borderline personality disorder. *Psychological Medicine*. 2014; 44(10): 2099–112.
- Sahlin H, Bjureberg J, Gratz KL, et al. Emotion regulation group therapy for deliberate self-harm: a multi-site evaluation in routine care using an uncontrolled open trial design. *BMJ Open.* 2017; 7(10): e016220.
- 117. Bjureberg J, Sahlin H, Hellner C, et al. Emotion regulation individual therapy for adolescents with nonsuicidal self-injury disorder: a feasibility study. BMC Psychiatry. 2017; 17(1): 411.
- Vigerland S, Lenhard F, Bonnert M, et al. Internet-delivered cognitive behavior therapy for children and adolescents: A systematic review and meta-analysis. *Clinical Psychology Review*. 2016; 50: 1–10.
- Andersson G, Cuijpers P, Carlbring P, Riper H, Hedman E. Guided Internet-based vs. face-to-face cognitive behavior therapy for psychiatric and somatic disorders: a systematic review and meta-analysis. *World Psychiatry*. 2014; 13(3): 288–95.
- 120. Andersson G, Titov N. Advantages and limitations of Internet-based interventions for common mental disorders. *World Psychiatry*. 2014; 13(1): 4–11.
- Rozental A, Boettcher J, Andersson G, Schmidt B, Carlbring P. Negative Effects of Internet Interventions: A Qualitative Content Analysis of Patients' Experiences with Treatments Delivered Online. *Cognitive Behaviour Therapy*. 2015; 44(3): 223– 36.
- 122. Gilmore AK, Ward-Ciesielski EF. Perceived risks and use of psychotherapy via telemedicine for patients at risk for suicide. *Journal of Telemedicine and Telecare*. 2019; 25(1): 59–63.
- 123. Wallin E, Maathz P, Parling T, Hursti T. Self-stigma and the intention to seek psychological help online compared to face-to-face. *Journal of Clinical Psychology*. 2018; 74(7): 1207-18.
- 124. Arshad U, Farhat ul A, Gauntlett J, Husain N, Chaudhry N, Taylor PJ. A Systematic Review of the Evidence Supporting Mobile- and Internet-Based Psychological Interventions For Self-Harm. *Suicide and Life-Threatening Behavior*. 2020; 50(1).

- 125. Bjureberg J, Sahlin H, Hedman-Lagerlöf E, et al. Extending research on Emotion Regulation Individual Therapy for Adolescents (ERITA) with nonsuicidal self-injury disorder: open pilot trial and mediation analysis of a novel online version. BMC Psychiatry. 2018; 18(1): 326.
- 126. Morthorst B, Olsen MH, Jakobsen JC, et al. Internet based intervention (Emotion Regulation Individual Therapy for Adolescents) as add-on to treatment as usual versus treatment as usual for non-suicidal self-injury in adolescent outpatients: The TEENS randomised feasibility trial. JCPP advances. 2022; 2(4): e12115–n/a.
- 127. Kazdin AE. Mediators and mechanisms of change in psychotherapy research. Annual review of clinical psychology. 2007; 3(1): 1-27.
- Kraemer HC, Wilson GT, Fairburn CG, Agras WS. Mediators and Moderators of Treatment Effects in Randomized Clinical Trials. *Archives of general psychiatry*. 2002; 59(10): 877–83.
- 129. Braun V, Clarke V. Using thematic analysis in psychology. *Qualitative Research in Psychology*. 2006; 3(2): 77–101.
- 130. Gratz KL, Latzman RD, Young J, et al. Deliberate Self-Harm Among Underserved Adolescents: The Moderating Roles of Gender, Race, and School-Level and Association With Borderline Personality Features. *Personality disorders: Theory, Research, and Treatment.* 2012; 3(1): 39–54.
- Gratz KL. Measurement of Deliberate Self-Harm: Preliminary Data on the Deliberate Self-Harm Inventory. *Journal of psychopathology and behavioral* assessment. 2001; 23(4): 253–63.
- 132. Miller AL, Rathus JH, Linehan MM, Swenson CR. *Dialectical Behavior Therapy with Suicidal Adolescents*. New York: Guilford Press; 2007.
- 133. Cwik MF, Walkup JT. Can randomized controlled trials be done with suicidal youths? *International review of psychiatry*. 2008; 20(2): 177–82.
- Cheong J, MacKinnon DP, Khoo ST. Investigation of Mediational Processes Using Parallel Process Latent Growth Curve Modeling. *Structural equation modeling*. 2003; 10(2): 238–62.
- Anckarsäter H, Lundström S, Kollberg L, et al. The Child and Adolescent Twin Study in Sweden (CATSS). *Twin research and human genetics*. 2011; 14(6): 495– 508.
- 136. Hansson SL, Svanstrom Rojvall A, Rastam M, Gillberg C, Gillberg C, Anckarsater H. Psychiatric telephone interview with parents for screening of childhood autism – tics, attention-deficit hyperactivity disorder and other comorbidities (A-TAC): Preliminary reliability and validity. *The British Journal of Psychiatry*. 2005; 187(3): 262–7.

- 137. Larson T, Anckarsäter H, Gillberg C, et al. The Autism Tics, AD/HD and other Comorbidities inventory (A-TAC): further validation of a telephone interview for epidemiological research. *BMC psychiatry*. 2010; 10(1): 1–.
- Kaufman J, Birmaher B, Brent D, et al. Schedule for Affective Disorders and Schizophrenia for School-Age Children-Present and Lifetime Version (K-SADS-PL): Initial Reliability and Validity Data. *Journal of the American Academy of Child and Adolescent Psychiatry*. 1997; 36(7): 980–8.
- Axelson D, Birmaher B, Zelazny J, Kaufman J, Gill M. K–SADS–PL 2009 Working Draft. Western Psychiatric Institute and Clinic: Advanced Center for Intervention and Services Research (ACISR) for Early Onset Mood and Anxiety Disorders; 2009.
- 140. Gould MS, Marrocco FA, Kleinman M, et al. Evaluating latrogenic Risk of Youth Suicide Screening Programs: A Randomized Controlled Trial. *JAMA : the journal of the American Medical Association*. 2005; 293(13): 1635–43.
- 141. Muehlenkamp JJ, Swenson LP, Batejan KL, Jarvi SM. Emotional and Behavioral Effects of Participating in an Online Study of Nonsuicidal Self-Injury: An Experimental Analysis. *Clinical psychological science*. 2015; 3(1): 26–37.
- McCauley E, Berk MS, Asarnow JR, et al. Efficacy of Dialectical Behavior Therapy for Adolescents at High Risk for Suicide: A Randomized Clinical Trial. JAMA Psychiatry. 2018; 75(8): 777–85.
- 143. Santamarina-Perez P, Mendez I, Singh MK, et al. Adapted Dialectical Behavior Therapy for Adolescents with a High Risk of Suicide in a Community Clinic: A Pragmatic Randomized Controlled Trial. Suicide & life-threatening behavior. 2020; 50(3): 652–67.
- Hofmann SG, Asnaani A, Vonk IJJ, Sawyer AT, Fang A. The Efficacy of Cognitive Behavioral Therapy: A Review of Meta-analyses. *Cognitive therapy and research*. 2012; 36(5): 427–40.
- 145. Griffiths H, Duffy F, Duffy L, et al. Efficacy of Mentalization-based group therapy for adolescents: the results of a pilot randomised controlled trial. *BMC psychiatry*. 2019; 19(1): 167–.
- 146. Mehlum L, Tørmoen AJ, Ramberg M, et al. Dialectical Behavior Therapy for Adolescents With Repeated Suicidal and Self-harming Behavior: A Randomized Trial. Journal of the American Academy of Child & Adolescent Psychiatry. 2014; 53(10): 1082–91.
- 147. Schäfer JÖ, Naumann E, Holmes EA, Tuschen–Caffier B, Samson AC. Emotion Regulation Strategies in Depressive and Anxiety Symptoms in Youth: A Meta– Analytic Review. *Journal of youth and adolescence*. 2017; 46(2): 261–76.

- 148. Pineda JPD, Dadds MRPD. Family Intervention for Adolescents With Suicidal Behavior: A Randomized Controlled Trial and Mediation Analysis. *Journal of the American Academy of Child and Adolescent Psychiatry*. 2013; 52(8): 851–62.
- 149. Rossouw T, Fonagy P. Mentalization-Based Treatment for Self-Harm in Adolescents: A Randomized Controlled Trial. *Journal of the American Academy of Child and Adolescent Psychiatry*. 2012; 51(12): 1304–13.e3.
- 150. Asarnow JR, Berk MS, Bedics J, et al. Dialectical Behavior Therapy for Suicidal Self-Harming Youth: Emotion Regulation, Mechanisms, and Mediators. *Journal of the American Academy of Child and Adolescent Psychiatry*. 2021; 60(9): 1105– 15.e4.
- 151. Gratz KL, Bardeen JR, Levy R, Dixon-Gordon KL, Tull MT. Mechanisms of change in an emotion regulation group therapy for deliberate self-harm among women with borderline personality disorder. *Behaviour Research and Therapy*. 2015; 65: 29– 35.
- 152. Cliffe B, Tingley J, Greenhalgh I, Stallard P. mHealth Interventions for Self-Harm: Scoping Review. *Journal of medical Internet research*. 2021; 23(4): e25140-e.
- Swift JK, Callahan JL, Cooper M, Parkin SR. The impact of accommodating client preference in psychotherapy: A meta-analysis. *Journal of clinical psychology*. 2018; 74(11): 1924–37.
- 154. McCashin D, Coyle D, O'Reilly G. Qualitative Synthesis of Young People's Experiences With Technology-Assisted Cognitive Behavioral Therapy: Systematic Review. Journal of Medical Internet Research. 2019; 21(11): e13540.
- 155. Biederman J, Mick E, Faraone SV. Age–Dependent Decline of Symptoms of Attention Deficit Hyperactivity Disorder: Impact of Remission Definition and Symptom Type. *The American journal of psychiatry*. 2000; 157(5): 816–8.
- Lockwood J, Daley D, Townsend E, Sayal K. Impulsivity and self-harm in adolescence: a systematic review. *European Child & Adolescent Psychiatry*. 2017; 26(4): 387-402.
- O'Connor RC, Kirtley OJ. The integrated motivational-volitional model of suicidal behaviour. *Philosophical Transactions of the Royal Society B: Biological Sciences*. 2018; 373(1754): 20170268-.
- Anestis MD, Soberay KA, Gutierrez PM, Hernández TD, Joiner TE. Reconsidering the Link Between Impulsivity and Suicidal Behavior. *Personality and social psychology review*. 2014; 18(4): 366–86.
- Kraemer HC, Frank E, Kupfer DJ. Moderators of Treatment Outcomes: Clinical, Research, and Policy Importance. JAMA : the journal of the American Medical Association. 2006; 296(10): 1286–9.

- 160. Schafer JL, Graham JW. Missing Data: Our View of the State of the Art. *Psychological methods.* 2002; 7(2): 147–77.
- 161. Kaess M, Koenig J, Bauer S, et al. Self-injury: Treatment, Assessment, Recovery (STAR): online intervention for adolescent non-suicidal self-injury – study protocol for a randomized controlled trial. *Trials*. 2019; 20(1): 425–.
- Clark DM. Realizing the Mass Public Benefit of Evidence-Based Psychological Therapies: The IAPT Program. *Annual review of clinical psychology*. 2018; 14(1): 159–83.
- 163. Ougrin D, Zundel T, Ng A, Banarsee R, Bottle A, Taylor E. Trial of Therapeutic Assessment in London: randomised controlled trial of Therapeutic Assessment versus standard psychosocial assessment in adolescents presenting with selfharm. Archives of disease in childhood. 2011; 96(2): 148–53.
- 164. Ougrin D, Kyriakopoulos M, Zundel T, Banarsee R, Stahl D, Taylor E. Adolescents With Suicidal and Nonsuicidal Self-Harm: Clinical Characteristics and Response to Therapeutic Assessment. *Psychological assessment*. 2012; 24(1): 11-20.
- 165. Aspvall K, Andersson E, Melin K, et al. Effect of an Internet-Delivered Stepped-Care Program vs In-Person Cognitive Behavioral Therapy on Obsessive-Compulsive Disorder Symptoms in Children and Adolescents: A Randomized Clinical Trial. JAMA : the journal of the American Medical Association. 2021; 325(18): 1863–73.
- 166. Sheppler CR, Edelmann AC, Firemark AJ, et al. Stepped care for suicide prevention in teens and young adults: Design and methods of a randomized controlled trial. *Contemporary clinical trials*. 2022; 123: 106959–.